

Asymmetries in relative clause comprehension in three European sign languages

Charlotte Hauser, Giorgia Zorzi, Valentina Aristodemo, Beatrice Giustolisi, Doriane Gras, Rita Sala, Jordina Sánchez Amat, Carlo Cecchetto, Caterina Donati

▶ To cite this version:

Charlotte Hauser, Giorgia Zorzi, Valentina Aristodemo, Beatrice Giustolisi, Doriane Gras, et al.. Asymmetries in relative clause comprehension in three European sign languages. Glossa: a journal of general linguistics (2016-2021), 2021, 6 (1), pp.72. 10.5334/gjgl.1454. hal-03246691

HAL Id: hal-03246691 https://hal.science/hal-03246691v1

Submitted on 20 Jul 2021

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

6

Asymmetries in relative clause comprehension in three European sign languages

CHARLOTTE HAUSER GIORGIA ZORZI VALENTINA ARISTODEMO BEATRICE GIUSTOLISI DORIANE GRAS RITA SALA [©] JORDINA SÁNCHEZ AMAT [©] CARLO CECCHETTO [©] CATERINA DONATI [©]

*Author affiliations can be found in the back matter of this article

RESEARCH

]U[ubiquity press

Abstract

Relativization is a robust subordinating type across languages, displaying important typological variability concerning the position of the nominal head that the relative clause modifies, and sign languages are no exception. It has been widely assumed since Keenan & Comrie (1977) that the subject position is more accessible to relativization than object and oblique positions. The main aim of this paper is to investigate the extension of this famous generalization both across modalities (sign as opposed to spoken languages) and across relativization typologies (internally as opposed to externally headed relatives), and to verify how it interacts with age of first language exposure. We here report the results of a sentence-to-picture matching task assessing the comprehension of subject and object relative clauses (RCs) in three sign languages: French Sign Language (LSF), Catalan Sign Language (LSC), and Italian Sign Language (LIS). The results are that object RCs are never easier to comprehend than subject RCs. Remarkably, this is independent from the type of relative clause (internally or externally headed). As for the impact of age of exposure, we found that native signers outperform non-native signers and that a delay in language exposure emphasizes the subject/object asymmetry. Our results introduce a new potential diagnostic for LF movement: the existence of a Subject Advantage in comprehension can be used as a reliable and measurable cue for the existence of long-distance dependencies, including covert ones.

CORRESPONDING AUTHOR: Charlotte Hauser

Université de Paris – LLF, CNRS. 8 place Paul Ricœur, 75013, Paris *charlotte.hauser@live.fr*

KEYWORDS:

Sign language; relative clauses; comprehension; Subject/ Object asymmetries; age of exposure; cross-linguistic and cross-modal typology

TO CITE THIS ARTICLE:

Hauser, Charlotte, Giorgia Zorzi, Valentina Aristodemo, Beatrice Giustolisi, Doriane Gras, Rita Sala, Jordina Sánchez Amat, Carlo Cecchetto and Caterina Donati. 2021. Asymmetries in relative clause comprehension in three European sign languages. *Glossa: a journal of general linguistics* 6(1): 72. 1–36. DOI: https://doi. org/10.5334/gjgl.1454

1 Introduction

Relativization is a very robust subordinating type across languages, displaying an important typological variability concerning the position of the nominal head the relative clause modifies (de Vries 2001). Sign languages are no exception, robustly displaying relative clauses and exhibiting the same typological variability that is attested in spoken languages.

It has been widely assumed since Keenan & Comrie's Accessibility Hierarchy (1977) that subject positions are more accessible than object positions in relativization. The main aim of this paper is to investigate the extension of this famous generalization both across modalities and across relativization typologies, and verify how it interacts with age of first language exposure (AoE). We here report the results of a sentence-to-picture matching task assessing the comprehension of subject and object relative clauses (RCs) in three sign languages: French Sign Language, LSF, Catalan Sign Language, LSC, and Italian Sign Language, LIS, reflecting some typological variability.

We have three main research questions. First: as predicted by the Accessibility hierarchy, externally headed relative clauses¹ display a clear subject/object asymmetry in spoken languages. We want to verify whether the same asymmetry holds in sign languages. LSF is especially relevant on this question, since it displays this type of relatives. Second, we want to investigate what happens with the comprehension of internally headed RCs, which have been poorly studied from this point of view, even in spoken languages. LIS and LSC both use internally headed relative clauses and provide data that are directly relevant for this question. Third, we aim at tracing the impact of age of first language exposure on the comprehension of relative clauses by comparing three groups of Deaf signers with different AoE to (sign) language.

Before entering into these questions and describing the way they are addressed in the structure of the paper, let us briefly remind how relatives vary typologically, and how sign languages enter this picture.

2 Main relativization strategies

Restrictive relative clauses are clauses modifying a nominal constituent, called the head. Irrespective of language modality, the main typological variation concerns the position of the head with respect to the relative clause modifying it.

Externally headed relative clauses, illustrated with the English example in (1) and the Turkish example in (2), display a clear separation between the clause and the nominal head (in bold), which is thus external to the relative clause.

- (1) the **cat** [$_{RC}$ that _ kicked the dog]
- (2) *Turkish* (Kornfilt 1997: 29)

[[_{RC -} geçen yaz ada-da ben-i gör-en] **kişi-ler**] _ last summer island-LOC I-ACC see-(SPart) person-PL 'the people who saw me on the island last summer'

While they are both externally headed, the relative clause in English (1) is postnominal, while the relative clause in Turkish (2) is prenominal, another important typological difference. In both cases, the relative clause displays a gap corresponding to the position where the head is interpreted. Both (1) and (2) are called *subject relative clauses* because the gap sits in the subject position of the embedded clause.²

In *internally headed relative clauses*, on the other hand, the noun modified by the relative clause is internal to the clause itself, which therefore contains no gap. This type of relativization

¹ We ignore here the important distinction between restrictive and appositive relative clauses, and only focus on restrictive relatives all along the paper.

² In some languages, such as Arabic for example (Aoun et al. 2010), externally headed RCs display a resumptive pronoun instead of the gap. This resumptive strategy will not be relevant here.

strategy is illustrated in (3) with a Japanese example. Here the head nominal sits in the object position within the relative clause, and (3) is thus an *object relative clause*.

Hauser et al. Glossa: a journal of general linguistics DOI: 10.5334/gjgl.1454

 (3) Japanese (adapted from Shimoyama 1999: 147)
 [_{RC} Taro-ga sara-no ue-ni keeki-o oita-no]-o [Taro-NOM plate-GEN on-LOC cake-ACC put-NM]-ACC
 'the piece of cake which Taro put on a plate'

Finally, RCs may vary as to their position within the main clause. While all the examples discussed so far present relative clauses in isolation, they can of course modify nominals that occupy various functions in the clause. The subject RC in (1) can for example modify a head that serves as the subject (4a) or object (4b) of the main clause.

(4) a. $[_{DP}$ The **cat** that kicked the dog] is black.

b. I punished $[_{DP}$ the **cat** that kicked the dog].

The languages under investigation in this paper, LSF, LIS and LSC, reflect the diversity of relativization strategies briefly outlined here, but in a modality, the signed modality, that also exhibits some important specificities.

2.1 Sign languages within linguistic research

Research into sign languages is a young field when compared to spoken language studies. With our study, we also aim at contributing to the understanding of the nature of sign languages with respect to spoken languages.

Over the past decades, an ever-growing body of research has shown that sign languages resort to mechanisms that are sufficiently similar to those observed in spoken languages as to allow crossmodal generalizations in all linguistic domains. At the same time, the visual modality does have an impact on language organization and articulation, and its unique features afforded by the modality should not be downplayed in linguistic descriptions. This is particularly clear when it comes to the grammatical use of the signing space, as reflected by the glossing conventions that have been elaborated specifically to report on sign languages. We here follow the conventions adopted in the SIGN-HUB grammars (*https://sign-hub.eu/grammar*), that we briefly introduce now to facilitate the reader of this paper.

First of all, signs are glossed using non-inflected English words in small capital letters. As for spatial information, it is encoded through the use of indices. When a nominal sign is articulated in a specific location in the signing space, this is signaled with an index letter. When a nominal sign is body anchored, it might still carry an index to indicate possible agreement or coreferential relations the sign might entertain. Pronominal pointing pronouns are referred through the gloss "IX", followed by an indexical number indicating person marking and/or a letter to indicate the specific location in the neutral signing space where they are articulated. Shared indices signal co-referring elements. The same conventions apply to directional verbs (i.e. verbs that display spatial agreement between a starting and an ending location, see Padden 1988) or any sign realized in a specific location in space. Finally, non-manual markers are linguistic forms that do not imply the use of the hand; they can include movement of the body or the eyebrows, eye-gaze direction, and any other facial expression. Non-manual markers and their spreading over manual signs are signaled through a line extending above the signs they co-occur with. These glossing conventions are exemplified in the following French Sign Language (LSF) sentence, in which "bt" refers to body turn, "re" to raised eyebrows and "rg" to right gaze.

(5) French Sign Language (Hauser 2019: 170)

 $\frac{\text{bt,re}}{\text{MARIE}_{a 3a} \text{SAY}_{1} \text{ POSS}_{3a} \text{ SISTER NICE}}$ 'Marie says to me that her sister is nice.'

2.2 The structure of this paper

The paper is organized as follows: Section 2 presents an overview of the different questions we are addressing, starting with what is known about biases in processing externally headed RCs,

followed by an outline of the puzzle of internally headed RCs (§2.2), and a brief overview of the relevance of age of first language exposure in syntactic competences (§2.3). Section 3 offers a description of how relativization works in the three sign languages under investigation, LSF (§3.1), LIS (§3.2) and LSC (§3.3). Section 4 presents our study on the comprehension of RCs in these three languages and Section 5 details its results, by presenting the analysis of accuracy (§5.1) and that of errors (§5.2). Section 6 provides a general discussion and Section 7 briefly concludes the paper.

Hauser et al. Glossa: a journal of general linguistics DOI: 10.5334/gjgl.1454

3 Dimensions of variation: The Accessibility Hierarchy across typology and the effects of first language exposure

A famous finding revealing asymmetries between different types of relatives is Keenan & Comrie's (1977) Accessibility Hierarchy, a scale elaborated on the basis of approximately 50 languages on accessibility to relativization according to the grammatical function of the RC head. It captures the fact that while all 50 languages allow relativization over a subject like in (1), repeated here as (6), fewer languages display object relative clauses like (7), and fewer and fewer languages allow relativization of further positions along the scale (see (8)).

(6) the **cat** [that __kicked the dog]

(7) the **cat** [that the dog kicked _]

(8)

Accessibility Hierarchy (Keenan & Comrie, 1977)				
Subject > Direct Object > Indirect Obj. > Oblique case, Object of Comparison				
> Genitive				

As they elaborate this scale, Keenan & Comrie make the hypothesis that this typological implication correlates with processing complexity, hence turning the Accessibility Hierarchy into a "predictor of psychological complexity" (Keenan & Comrie 1977: 61). Subject relative clauses would be easier and faster to process and acquired earlier than object relative clauses, and the same would hold for further positions along the scale. Their hypothesis was first supported by Keenan & Hawkins' (1974/1987) study on language acquisition, through a sentence repetition task, in which the performance of 10-year-old English children statistically correlated with the position of the relative head along the Accessibility Hierarchy.

This phenomenon, referred to as the "Subject Advantage", holds across methods and languages. As for English, subject RCs are produced at an earlier age in acquisition and in higher proportion than object RCs (Diessel 2004). Adults read object RCs significantly slower than subject RCs and obtain lower accuracy scores in comprehension questions (King & Just 1991). A similar Subject Advantage in the form of longer fixations on object RCs compared to subject RCs was also captured in eye-tracking studies (Traxler et al. 2005) and in Electroencephalogram (EEG) studies that associated complexity to a higher electrical signal of the Left Anterior Negativity (LAN) (Müller, King, & Kutas 1997). The Subject Advantage was also confirmed in functional Magnetic Resonance Imagery (fMRI) studies (Constable et al. 2004) and Positron Emission Tomography (PET) studies (Stromswold et al. 1996). These results have been replicated in French (Baudiffier et al. 1990), German (Mecklinger et al. 1995) and many other languages.

As to how to explain such asymmetry, there is no general consensus, with proposals pointing at resource-based effects related to structural distance (Frazier 1987; Hawkins 1999), intervention (Friedmann et al. 2009), linear distance (e.g. King & Just 1991; Gibson 2000) or, canonical order effects (Diessel & Tomasello 2005), distribution-based effects (e.g. Mak et al. 2006), and prominence-factors (van Valin & Wilkins 1996). These different accounts make different predictions about the cross-linguistic extension of the Subject Advantage. While they all predict a Subject Advantage in SVO languages with postnominal relatives, like English, they differ in their predictions concerning other word orders and other relativization strategies.

The structural distance hypothesis (Hawkins 1999) predicts the asymmetry to hold across word orders and relativization strategies, at least as far as externally headed relative clauses are

concerned. As for postnominal RCs, as in English, the idea is that the complexity captured by the Accessibility Hierarchy derives from the fact that speakers need to keep the head of the RC in their memory until they encounter the gap position. Since an object gap, as in (7), is structurally further away from the head than a subject gap like (6), an object RC is harder to process than a subject RC. The same asymmetry is predicted to arise with prenominal RCs, where the speaker needs to wait to find the head of the RC once they encounter a gap. Structurally (but not linearly), the gap of an object RC (as 10 in Turkish) is more deeply embedded than that of a subject RC (cf. 2 repeated as 9 below).

Hauser et al. Glossa: a journal of general linguistics DOI: 10.5334/gjgl.1454

- (9) Subject relative clause in Turkish (Kornfilt, 1997: 29)
 [[_{RC _} geçen yaz ada-da [_{VP} ben-i gör-en]] kişi-ler] _ last summer island-LOC I-ACC see-(SPART) person-PL
 'the people who saw me on the island last summer'
- (10) Object relative clause in Turkish (Kornfilt, 1997: 29)
 [[_{RC} pro geçen yaz ada-da [_{VP} _ gör-düg-üm]] kişi-ler] pro last summer island-LOC _ see-OPART-1.SG person-PL
 'the people who(m) I saw on the island last summer'

The same prediction of a universal subject advantage holds for other structural hypotheses, such as Relativized Minimality and Intervention (Friedman et al. 2009), given that by definition a subject will invariably intervene structurally between an object gap and its antecedent, no matter their respective linear positions.

Other accounts that are based on linear factors predict the Subject Advantage to hold only with some word orders and some types of relativization strategies. In particular, accounts based on linear distance predict that languages with prenominal relative clauses should exhibit an object advantage. To go back to the Turkish examples above, the subject gap is linearly further away from the head in (9) than the object gap in (10). If what counts is linear distance, an object advantage is predicted to hold.

Most studies³ point towards a universal Subject Advantage, holding both with postnominal and with prenominal RCs (see results on Korean by Kwon Nayoung et al. 2013; Kwon Nayoung et al. 2010; O'Grady, Lee, & Choo 2003; Slobin & Zimmer 1986; but also on Japanese by Miyamoto & Nakamura 2013; Turkish by Özge, Marinis, & Zeyrek 2009; on Cantonese by Huang Jiaying & Donati 2019; on Wenzhounese by Hu Shenai, Cecchetto & Guasti 2018; on Mandarin Chinese by Lin Chien-Jer & Bever 2006; Wu Fuyun et al. 2009; Lau Elaine 2016 and Jäger et al. 2015 a.o.).

Across studies, the Subject Advantage has been shown to be affected and modulated by a number of factors (Vasishth & Lewis 2006). Traxler et al. (2005) and Mak et al. (2006) show that inanimate heads in object RCs drastically reduce the Subject Advantage in English. Similarly, Friedmann & Novogrodsky (2004) show that children easily understand object RCs if the relative clause has a pronominal subject, as in example (10).

(10) I prefer the **princess** [_{RC} that you are drawing ___].

Other factors, such as saliency (Roland et al. 2012; Hale 2016), appear to also have an impact. For this reason, when conducting cross-linguistic studies, it is important to use comparable experimental settings in order to control for the factors that are manipulated. In the experiments that we report about in this article, we will only focus on the syntactic position of the head (subject vs object), and will keep other parameters constant.

Another important factor of variation that might affect the comprehension of a RC concerns its position within the main clause. To clarify, the RC in (10) modifies an object NP, while the same RC in (11) modifies a subject NP.

(11) The **princess** [_{RC} that you draw ___] is strong and powerful.

³ See Gibson (2000), Qiao, Shen, & Forster (2012) or Chen & Shirai (2015) a.o. for data suggesting the existence of an object advantage in prenominal RCs.

In order to avoid any possible effect of this variable, all the RCs in our experiments modify an object NP.

Hauser et al. Glossa: a journal of general linguistics DOI: 10.5334/gjgl.1454

The vast literature on asymmetries in the comprehension, processing, production and acquisition of relative clauses has at least two limitations. One concerns the type of languages investigated, which does not include the sign modality (or barely so). The other concerns the relativization strategy, which is mostly that of (prenominal or postnominal) externally headed relative clauses. One aim of this paper is to contribute to filling these two gaps. As we shall see, LSF displays externally headed RCs, while LIS and LSC both make use of internally headed RCs. If the Accessibility Hierarchy based on the study of spoken languages is valid across modalities, the predictions are straightforward for LSF: since it presents externally headed RCs, we should find a clear Subject Advantage in relative clause comprehension. For LIS and LSC, on the other hand, the predictions are less clear since, to this date, there is close to no research carried out on the Accessibility Hierarchy with internally headed RCs.

3.1 The puzzle of internally headed relative clauses

Internally headed RCs are clauses involving no gap, which are however nominal in nature (Cole 1987; Kayne 1994; Keenan & Comrie 1977; or Lehmann 1986). They can even show nominal inflection, as in Japanese, where a case marker (accusative) is attached to the internally headed relative clause in (12).

(12) Japanese (Shimoyama, 1999: 147)

Yoko-wa [_{RC} Taro-ga sara-no ue-ni **keeki**-o oita-no]-o tabeta. Yoko-Top [Taro-NOM plate-GEN on-LOC cake-ACC put-NM]-ACC ate 'Yoko ate the piece of cake which Taro put on a plate.'

Internally headed RCs are analyzed by Kayne (1994), Cole (1987) and Culy (1991), among many others, as the mirror image of externally headed RCs: just like them, they modify a head noun (*keeki*, 'cake in 12) that is shared semantically and syntactically between the main and the relative clause. The only difference between the two constructions from a structural perspective is the overt position of the head, which is either external or internal. For proponents of raising theories, like Cecchetto & Donati (2015), De Vries (2001) or Bianchi (2000), internally headed RCs involve the *covert* movement of the relative head while externally headed RCs involve the *overt* movement of the relative head. For proponents of head-matching theories, like Sauerland (2003) or, more recently, Cinque (2019), internally headed RCs and externally headed RCs are both base-generated with two head nouns, one internal and one external, with one of them deleted at PF.

In both families of analyses, there is no gap within internally headed RCs at least in overt syntax. If the Subject Advantage attested for externally headed RCs is due to the structural distance between the head noun *and the gap* (possibly because the identification of a null category is a complicated business), we would predict internally headed RCs to display no difference in comprehension between subject RCs and object RCs. However, both families of analyses posit that internally headed RCs involve an abstract dependency between two positions. Is comprehension affected by the abstract dependency involved in internally headed RCs? If it is, we would predict a Subject Advantage in internally headed RCs as in externally headed RCs.

To this date, very little research has been conducted with these questions in mind, and all come from the domain of L1 and L2 acquisition studies. Jeon Seon & Hae-Young Kim (2007) analyzed the elicited production of 40 L2 learners of Korean, a language displaying both internally headed RCs and externally headed RCs. The aim was to verify whether there is an asymmetry in the acquisition of Subject RCs and Object RCs in head-internal and head-external constructions. What they found was a clear Subject Advantage for externally headed RCs, with L2 learners producing Subject RCs more frequently and more accurately (91.1% of correct answer) than Object RCs (83.7%), but a less clear pattern for internally headed RCs. The authors do not report the accuracy scores, but note that Object RCs are slightly more frequent (6%) than Subject RCs (4. 8%). Similarly, Hu Shenai, Cecchetto & Guasti (2018) made an elicited production study of RCs in Wenzhounese, a language that allows both internally headed RCs, but an object preference in internally headed RCs.

Hauser et al. Glossa: a journal of general linguistics DOI: 10.5334/gjgl.1454 7

The order of acquisition of internally headed RCs and externally headed RCs itself seems to be different, as shown by a longitudinal study on three Korean children's L1 acquisition conducted by Kim Youn-Joo (1987), who found that Korean children start to produce headless relative clauses around 2 y.o., before producing internally headed RCs, and that externally headed RCs come last. This pattern is also found by Lee Awen Ock's (1991) investigation of the spontaneous production of 36 Korean children, with a mean age of acquisition of internally headed RCs around 27 months, and around 36 months for externally headed RCs. The same pattern is confirmed by Jeon Seon & Hae-Young Kim (2007) for L2 learners of Korean.

These scattered results, although interesting, are not sufficient to draw any conclusions regarding the Accessibility Hierarchy and its predictive power when it comes to internally headed RCs. It is however interesting to note that there seems to be a difference across RC types, and some studies suggest that no Subject advantage or even an opposite Object advantage might hold for internally headed RCs. In the narrow light of these studies, the predictions regarding the Accessibility Hierarchy and the Subject Advantage (if any) when it comes to internally headed RCs are rather blurred. If difficulty arises from structural distance or intervention, there are two possibilities: if what matters is the structural distance between the head noun and the edge of the relative clause where a silent copy is interpreted, we expect to observe a Subject Advantage in internally headed RCs similar to what is observed in externally headed RCs. If on the other hand what matters is that in internally headed RCs there is no gap to fill, we should expect no asymmetry between object RCs and subject RCs.

Additionally, if internally headed RCs are really easier to understand than externally headed RCs, as the studies reported above appear to suggest, this might make the detection of a Subject Advantage more difficult with the former, with participants being too accurate to observe any difference. Yet, on the other hand, the head noun is not easy to identify in internally headed RCs, since it is not singled out by its position. This might involve an extra burden that is not required in the processing of externally headed RCs.

Our study is likely to be able to provide some answers to these questions since both LSC and LIS exploit internally headed RCs. Besides, as we shall see below, Deaf signers include very heterogeneous populations, where differences in AoE might make visible even small processing biases.

3.2 Age of first exposure to a sign language

For the vast majority of hearing infants, language exposure starts already within the womb (see DeCasper & Spence 1986, or Gervain et al. 2008 on that matter) and language deprivation is restricted to exceptional cases of severe child abuse. But for the vast majority of deaf infants born in a hearing family, a delay in language exposure occurs, which is due to an incompatibility between the perceptual abilities of the child and the surrounding language. Even with early intervention through hearing aids or cochlear implants, language access is delayed if not provided from birth in a fully accessible way, which is the visual modality (see Hall et al. 2019 or Humphries et al. 2016 a.o.).

As a result, Deaf⁴ signing communities are very heterogeneous: few individuals have signing parents/relatives and are exposed to a sign language from birth (i.e. they are native signers in the narrow sense), whereas the vast majority, having non-signing parents, are exposed only to a spoken language in their early life and discover sign language later. This group is composed of people with quite different linguistic profiles, since outcomes of exposure to spoken language through hearing aids only are very variable across individuals, and late exposure to a sign language does not ensure full language acquisition (see Humphries et al. 2016).

The long-lasting effects of this delayed acquisition in adulthood have been discussed in a number of studies focusing mainly on American Sign Language (ASL) and British Sign Language (BSL). In 1993, Mayberry reported that native signers are better at recalling complex ASL sentences than late learners and that, more broadly, the age of ASL exposure is correlated to task performance. This result was confirmed in Boudreault & Mayberry's (2006) experiment,

⁴ We follow here the standard practice of capitalizing *Deaf* when we refer to individuals who are deaf and identify culturally as belonging to a community, while we keep the simple *deaf* when referring to their auditory condition.

and in its BSL version (Cormier et al. 2012), where participants had a limited time to determine the grammaticality of a set of sentences they were presented with. Stimuli ranged from simple declarative sentences to more complex ones like interrogatives or relative clauses, and participants' performance decreased as their age of exposure to a sign language increased. At the morpho-syntactic level, Emmorey et al. (1995) highlight the difference in perceiving agreement errors in ASL sentences when comparing native and non-native signers. These results also extend to the morpho-phonological level, with native signers being quicker and more accurate in distinguishing between words and pseudo-words (Emmorey & Corina 1990). It is thus at all levels of the linguistic signal that effects of a delayed sign language exposure can be traced.

These effects have been mainly found in ASL and BSL signers even though their home countries are considered as very advanced in the recognition and diffusion of sign languages: the United States even host the only fully-ASL accessible university, and the United Kingdom is a pioneer in the development and use of sign language in linguistic assessment (see Haug 2005 for a review). This narrow focus makes it all the more important to investigate adult linguistic performance in European sign languages, where deeply rooted public policies favoring oralist education⁵ have largely impeded the recognition of sign languages and their importance in individuals' cognitive and linguistic development. It is thus essential to investigate the consequences of a delayed exposure to sign language to better inform policy makers and support Deaf people's efforts in this direction.

4 Relative clauses in sign languages

The past 50 years have been rich in research documenting sign language relativization strategies. To date, studies have been carried out on ASL (Liddell 1978, Liddell 1980, Fontana 1990; Miller 1990; Galloway 2011); LIS (Cecchetto et al. 2006, Branchini & Donati 2009, Branchini 2014), Brazilian Sign Language (Libras, Nunes & de Quadros 2005), German SL (DGS, Pfau & Steinbach 2005, Happ & Vorköper 2006), Japanese SL (JSL, Penner et al. 2019), Hong Kong SL (HKSL, Tang Gladys & Prudence Lau 2012; Li 2013), Turkish SL (TiD, Kubus 2011); LSC (Mosella Sanz 2012), and LSF (Hauser 2016; 2019; Hauser & Geraci 2017). These studies demonstrate that sign languages display as much typological variation as found in spoken languages, with some displaying internally headed RCs (ASL, LIS or LSC a.o.), others pre-nominal externally headed RCs (LSF, ASL, DGS).

In the vast majority of sign languages, relative clauses are signaled through the use of specific manual and/or non-manual markers, the only exception being JSL, where according to Penner et al. (2019) there is no marking whatsoever signaling relativization. For example, ASL RCs are signaled through backward head tilt, raised eyebrows, and tensed upper lip (Wilbur, 2017). A manual sign dedicated to relativization can also be used. Its nature varies across languages, from relative pronouns (13), to complementizers (14) or nominalizers (15). The relevant manual signs are underlined in the following examples.

(13) *LSF* (adapted from Hauser & Geraci, 2017: 20)

 $\frac{\text{rel}}{\text{IX}_{1} \text{ PREFER } \mathbf{DOG}_{1} \left[_{\text{RC}} \underbrace{\text{PI}_{1}}_{\text{I}} \text{ MAN PET} \right]$ 'I prefer the dog that the man pets.'

(14) ASL (adapted from Liddell, 1978: 85)

¹ASK₃ GIVE₁ **DOG** [$_{RC}$ URSULA KICK <u>THAT</u>_C] 'I asked him to give me the dog that Ursula kicked.' 8

⁵ Starting from the International Congress for the Education of the Deaf (ICED) held in Milan in 1880 most Western and European countries based deaf education on so-called "oralism", that is exclusive focus on oral language training by using lip reading and mimicking mouth shapes and breathing patterns of speech. Many European countries have made important steps towards a more inclusive education using sign language, but the oralist tradition is still alive in most language policies towards the deaf (see Lane 1979).

(15) *LSC* (adapted from Mosella Sanz, 2012)

rel

 $[_{\rm RC}$ YESTERDAY **MAN** COME <u>SAME</u>] POSS₁ FRIEND 'The man who came yesterday is my friend.'

Regarding comprehension of RCs in sign languages, however, there is close to no research on the topic. Very recently, Hauser & Pozniak (2019) showed that LSF presents a clear Subject Advantage, using an eye-tracking visual world preference paradigm design. Being the first study of this kind, however, its results need to be replicated and expanded, especially with respect to effects of age of first language exposure on participants' accuracy.

In order to start filling in this gap, we chose to analyze the comprehension of relative clauses in LSF, LIS and LSC, three European sign languages that are typologically diverse. Here we briefly report the main features of relativization in the three languages, as they have been described in the literature.

4.1 Relative clauses in French Sign Language

LSF allows both SVO and SOV word orders with preference for one or the other varying across individuals (Hauser 2019). It mostly presents externally headed RCs which can be marked in three different ways: through non-manual markers only, with the relative pronoun PI or with the classifier PERSON-CL when the relative clause head is a human entity (Hauser & Geraci 2017; Hauser 2019). All three strategies come with a specific set of non-manual markers which can optionally spread over the head noun as well, glossed as "rel": raised eyebrows, body and head turned towards the locus of the relative pronoun or the head noun with which it agrees in space (see *Figure 1*).



a. Non-manual markers on the head noun (VET).



b. Non-manual markers on the relative pronoun PERSON-CL.



c. Non-manual markers on the relative pronoun PL

Figure 1 Relativization

markers in LSF, from Hauser (2019).

In the present study, we focused on the PI-strategy as it is the only strategy that receives an unambiguous restrictive interpretation with any type of head, whether human or not. The sign PI is glossed after its mouthing component since signers co-articulate the syllable [pi] when signing it. It starts with a hand configuration similar to a rounded pinch (F-shape but with non-selected fingers bent) and ends as a pointing sign (similar to the French gesture called "pichenette") agreeing in space with the head noun. An example of subject and object relative clauses in LSF is given in (16) and (17).

(16) Subject relative clause, LSF (Hauser 2019: 58)

IX₁ PREFER **MAN**_i [$_{RC}$ PI_i __ PET DOG] 'I prefer the man who pets the dog.'

(17) *Object relative clause, LSF* (Hauser 2019: 59)

IX₁ PREFER **DOG**_i [_{RC} PI_i MAN PET_] 'I prefer the dog which the man pets.'

To avoid variation, all the stimuli we built for our study follow the structure described in the literature, with an SVO order and the relative clause in its base position (Hauser 2019). An

example from our dataset is provided for subject relative clauses (see 18 and *Figure 2*) and for object relative clauses (19 and *Figure 3*).

Hauser et al. Glossa: a journal of general linguistics DOI: 10.5334/gjgl.1454

(18) Subject relative clause, LSF

 $\frac{\text{rel}}{\text{HAVE-TO CHOOSE GIRAFFE}_{i}} \begin{bmatrix} \text{RC PI}_{i} & \text{LICK}_{j} & \text{COW}_{j} \end{bmatrix}$ '(You) have to choose the giraffe that licks the cow.'



Figure 2 Screen captures of the sentence in (18), a subject relative in LSF.

Figure 3 Screen captures of the sentence in (19), an object

relative in LSF.

(19) *Object relative clause, LSF*

HAVE-TO CHOOSE **GIRAFFE**_i [$_{RC}$ PI_i COW_{j j}LICK_i] '(You) have to choose the giraffe that the cow licks.'



4.2 Relative clauses in Italian Sign Language

LIS is an SOV language. Its most common relativization strategy is to use a fronted internally headed RC⁶ marked on its right edge by the relative determiner PE (*Figure 4*). The relative marker PE agrees in location with the head noun. The whole relative clause is marked by a set of non-manual markers glossed as "rel" described as "raised eyebrows, a specific tension of the eyes and upper cheeks, and a forward head nod co-occurring with the sign PE" (Branchini 2014: 189).



Figure 4 The sign PE in LIS, from Hauser (2019).

6 See Cecchetto, Geraci & Zucchi (2006) for an alternative analysis in terms of correlative construction.

10

An example of a subject (20) and an object (21) RC in LIS is provided below.

(20) Subject relative clause, LIS (Branchini 2014: 191)

 $\frac{\text{rel}}{\left[_{RC} \mathbf{DOG}_{i} \text{ IX}_{3i} \text{ EAT A-LOT PE}_{i}\right] \text{ DOCTOR VET BRING}}$ 'I took the dog that eats a lot to the vet.'

(21) *Object relative clause, LIS* (Branchini 2014: 192)

 $\frac{\text{rel}}{\left[_{\text{RC}} \text{ YESTERDAY } \mathbf{DOG}_{i} \text{ FIND } \text{PE}_{i} \right] \text{ PAOLO}_{k} \text{ IX}_{k} \text{ WASH}}$ 'Paolo washed the dog that I found yesterday.'

For the present study, in order to avoid variation across stimuli, we generalized the use of a classifier handshape $(CL)^7$ that directly follows the object within the RC (see Benedicto & Brentari 2004, for a description of a similar phenomenon in ASL), as this was strongly preferred by most of our LIS informants. Example (22) and *Figure 5* illustrate a subject RC and sentence (24) along with *Figure 6* present an object RC using this classifier, both examples are taken from our data set.

(22) Subject relative clause, LIS

 $\begin{bmatrix} 1 \\ R_{C} & \text{GIRAFFE}_{k} & COW_{j} & CL_{j} & LICK_{j} & PE_{k} \end{bmatrix}$ CLICK 'Click on the giraffe that licks the cow.'



(23) Object relative clause, LIS

 $\frac{\text{rel}}{\left[_{\text{RC}} \text{ GIRAFFE}_k \text{ CL}_k \text{ COW}_j \text{ }_j \text{LICK}_k \text{ PE}_k\right] \text{ CLICK}}$ 'Click on the giraffe that the cow licks.'



Figure 6 Screen captures of the sentence (23) in LIS, an object relative clause.

4.3 Relative clauses in Catalan Sign Language

Catalan Sign Language (LSC) is also an SOV language having fronted internally headed RCs as the most common strategy for relativization. Here, relatives are signaled either through the use of non-manual markers only, or through the use of a sign glossed as MATEIX⁸ in the literature, meaning 'same'. This sign is realized on the shoulder with an arc movement (*Figure 7*). Despite this sign being body-anchored and hence not displaying any agreement, Mosella Sanz (2012)

Hauser et al. Glossa: a journal of general linguistics DOI: 10.5334/gjgl.1454 11

Figure 5 Screen captures of the sentence in (22), a subject relative clause in LIS.

⁷ Classifier are signs that can refer to a given class of entities (like vehicles, persons ...) and which are used to give spatial, shape and/or motion descriptions. Classifier handshapes are drawn from a closed set and differ from one sign language to another. For a full description see Quer et al. (2017).

⁸ From this point, this sign will be glossed as SAME.

notes that the head noun is signaled through body leaning (i.e. the signer leans in the direction of the head locus). The whole relative clause is marked through raised eyebrows, squinted eyes and slightly raised shoulders (Mosella Sanz 2012).

Hauser et al. Glossa: a journal of general linguistics DOI: 10.5334/gjgl.1454



Figure 7 The sign MATEIX/SAME in LSC, from Barberá (2020).

According to the author, SAME can appear either at the left (see 25, in an object relative clause) periphery of the relative clause or at the right periphery of the relative clause (as in 24, a subject relative clause) without affecting the meaning.

(24) Subject relative clause, LSC (adapted from Mosella Sanz 2012: 186)

 $\frac{\text{rel}}{\left[_{\text{RC}}\text{NOW GIRL IX}_{3}\text{ AREA CITY LIVE IX}_{\text{LOC}}\text{ SAME}\right] \text{ YESTERDAY ONION }_{3}\text{GIFT}_{1}$ 'Yesterday we gave onions to the girl who now lives in the city.'

(25) *Object relative clauses, LSC* (adapted from Mosella Sanz 2012: 186)

 $\frac{\text{rel}}{\left[_{\text{RC}}\text{ SAME } \textbf{MUSHROOM } \text{COLLECT}\right] \text{ BASKET } \text{CL-PUT-IN-BASKET } \text{'I will put the mushrooms I found in the basket.'}$

The strategy with the relative marker on the left was preferred by our informants, therefore we adopted it consistently across stimuli. We also minimized variation by always fronting the relative clause before the main clause. An example from our dataset is provided for subject relative clauses (see 26 and *Figure 8*) and for object relative clauses (27 and *Figure 9*).

(26) Subject relative clause, LSC

 $\label{eq:constraint} \underbrace{\frac{\text{rel}}{\sum_{\text{RC}} \text{SAME } \textbf{GIRAFFE}_{i \ i} \text{LICK}_k \text{COW}_k] \ \text{IX}_2 \text{CLICK}}_k \text{Click on the giraffe that is licking the cow.'}$



Figure 8 Screen captures of the sentence in (26), a subject relative clause in LSC.

 $\frac{\text{rel}}{\left[_{\text{RC}} \text{ SAME } \text{GIRAFFE}_{i} \text{ COW}_{k \ k} \text{LICK}_{i}\right] \text{ IX}_{2} \text{ CLICK}}$ 'Click on the giraffe that the cow licks.'



5 The present study

This section presents our experiments⁹ on RCs comprehension. Remember we had three main goals. The first goal was to investigate whether the Subject Advantage extends across modalities. The LSF experiment is directly relevant for this goal, since LSF displays externally headed postnominal RCs, which invariably exhibit the strongest Subject Advantage in the spoken modality. We thus expect to find a clear Subject Advantage in RC comprehension in this language.

The second goal was to investigate what happens with the comprehension of internally headed RCs, which have been poorly studied from this point of view. LIS and LSC both exhibit internally headed RCs and can provide data that are directly relevant for this question. There are few studies on this topic and they are partially contradictory, therefore it is difficult to have clear expectations. Even looking at theories that have been proposed to explain a Subject Advantage does not offer an unambiguous base to make a prediction.

No Subject Advantage is expected in internally headed RCs if the crucial factor that determines the Subject Advantage is ultimately related to the presence of a gap, say because the subject gap is easier to retain in the memory buffer, being closer to the relative clause head (either linearly or structurally). However, several researchers have proposed that internally headed RCs are only superficially different from externally headed RCs, since the same long-distance movement operation is involved in both, the only difference being that in externally headed RCs the movement is visible while in internally headed RC it applies at LF. If this analysis is correct, and if distance impacts syntactic movement even when the latter takes place at LF, a Subject Advantage might arise in internally headed RCs as well.

To further muddle the picture, internally headed RCs involve another factor of complexity, independent from the subject RC vs object RC contrast, namely the fact that the head is not explicitly marked by occupying a specific position. We expect this factor to impact our results and to reinforce any other bias that we might observe.

The third goal concerns the investigation of the impact of age of first exposure (AoE) on the comprehension of RCs. We expect native signers to outperform early and late signers, and that more complex sentences should be particularly challenging for non-native signers in general and late signers in particular. This means we expect the Subject Advantage to be stronger in non-natives in LSF, and whatever bias we observe with internally headed RCs to be stronger in non-natives in LIS and LSC.

5.1 Design and procedure

For each sign language, we developed a sentence-to-picture matching task based on Friedmann & Novogrodsky (2004). In each picture, three characters are displayed: two of them are identical and are either performing an action or undergoing that action with respect to a third different character standing between them (*Figure 10*).

Hauser et al. Glossa: a journal of general linguistics DOI: 10.5334/gjgl.1454

Figure 9 Screen captures of the sentence in (27), an object relative clause in LSC.

13

⁹ All the video items used in the experiments and the scripts for the analyses can be found in the following OSF repository: *https://osf.io/5bdu2/?view_only* = 683792bb06d2410facb49037a02a85b5. Images can be consulted by contacting the corresponding author.



Hauser et al. Glossa: a journal of general linguistics DOI: 10.5334/gjgl.1454

Figure 10 Three characters, one giraffe licks the cow, and the other one is licked by the cow.

Participants are asked to choose one of the characters based on a sentence containing a relative clause. The head of the relative clause invariably corresponds to one of the matching characters (the giraffes in *Figure 10*), and the RC is either a subject RC (i.e. 'Choose the giraffe that licks the cow') or an object RC (i.e. 'Choose the giraffe that the cow licks'). Some of the pictures were used as fillers, hence associated with a simpler request not including any relative clause (i.e. 'Choose the black and white animal'). The function of the fillers was three-fold. 1) As fillers were simple, participants did not need to keep the same level of concentration throughout the whole experiment and could rest. 2) Performance in fillers was used as a baseline measure of attention and overall comprehension. Participants who were lower than 75% correct in the fillers were excluded. 3) Fillers provided some trials targeting the middle character. Since the experimental design tends to bias participants to always click on the left or right characters, half of the fillers were targeting the middle character while the other half were targeting side characters.¹⁰

The protocol had a within-participant design, such that the experiment contained two blocks, administered to each participant in two separate sessions or even on separate days, which were the mirror image of each other: if a picture was presented with a subject RC stimulus in the first block, it was then presented with an object RC stimulus in the second block, so that the same picture was never displayed within the same block. Within a single block, half of the items were subject RCs and the other half were object RCs. The same fillers were used across the two blocks. Items and fillers were fully randomized. All the sentences used in the experiments have been elicited with native informants of the investigated sign language. Non-manual markers typical of relativization strategies in each language (see Section 3) are not indicated in the gloss.

The test always started by a video presenting the instructions in the relevant sign language, followed by a short training phase. At the end of the training phase, participants could ask clarification questions. There were minor variations in the general design and procedures across sign languages that we detail now.

5.1.1 LSF

For the LSF experiment, each block contained 42 trials (14 subject RCs, 14 object RCs, 14 fillers) preceded by 3 training trials – one subject RC, one object RC and one filler. The full list of items is given in the Appendix. The duration of each testing session was around 20-25 minutes and was embedded within a larger testing session (\sim 1h30) during which the participants performed other lexical and syntactic tests.

10 The experiment in LSC presented some minor differences concerning fillers. We detail them in Section 4.1.3.

Participants were left alone in the experiment room to perform the test on their own. They were sitting approximately 45 cm away from the screen (4:3 display, 22" screen) and answered using a mouse by clicking on the selected character.

For each trial, participants saw first the stimulus video, automatically followed by the picture on which to click to provide their answer (see *Figure 11*). The stimulus video always started by a small context introducing the characters present in the video (see 28), followed by a request embedding a subject RC (28a), or an object RC (28b) or, if it was a filler, a simpler sentence (28c). Notice that the context did not give any indication regarding the spatial disposition of the various characters, while the request embedding the RC displayed spatial location associated to the arguments of the clause. This spatial information did not always match the spatial arrangement of the characters in the picture (meaning that the left character on the picture was not always signed on the left side of the signer) since we did not want participants to be biased in their answers by this parameter. To avoid participants interpreting spatial information topographically nonetheless, the picture was only made visible after the end of the recorded sentence.

(28) Context, LSF

HERE THREE CHARACTER. TWO GIRAFFE ONE COW... 'There are three characters, two giraffes and a cow.'

- a. Subject relative clause
 ... HAVE-TO CHOOSE GIRAFFE_i [_{RC} PI_i ____ iLICK_k COW_k].
 '(You) have to choose the giraffe that licks the cow.'
- b. Object relative clause
 ... HAVE-TO CHOOSE GIRAFFE, [_{RC} PI, COW, kLICK,].
 '(You) have to choose the giraffe that the cow licks.'
- c. Filler

... HAVE-TO CHOOSE ANIMAL WHITE BLACK. '(You) have to choose the black and white animal.'¹¹

Experiments took place at the Université de Paris and at the Université de Nantes.



Hauser et al. Glossa: a journal of general linguistics DOI: 10.5334/gjgl.1454

Figure 11 Illustration of an item in LSF.

11 This is just an illustration. In the experiment, pictures used for target items are never used for fillers (but they have the same structure, with two identical characters interacting with a third, different one).

5.1.2 LIS

For the LIS experiment, each block contained 32 trials (10 subject RCs, 10 object RCs, 12 fillers) preceded by 3 training items (one example of each type of trial). The full list of items is given in the Appendix. The duration of each block was around 15 minutes and it was embedded in a larger testing session (1h) in which participants were undertaking other lexical and syntactic tests.

Participants were sitting in a bright room in front of a laptop with a 15" screen and answered using the touch screen. For each trial, participants first saw the stimulus video (no context provided) accompanied by the corresponding small-size picture in the bottom left corner of the screen, and then the picture alone in the middle of the screen on which they could touch the correct character to provide their answer (see *Figure 12*). In half of the items the arrangement in space of the various arguments in the request sentence matched that of the various characters in the picture; in the other half, there was a clear mismatch. This pattern was adopted to avoid that the simultaneous presentation of a miniature of the picture and the stimulus video induce a topographical interpretation of the use of the signing space in the latter. The stimulus video consisted of a request embedding a subject RC (29a), an object RC (29b) or, if the item was a filler, a simpler sentence (29c) targeting the middle character.

- (29) LIS
 - a. Subject relative clause $\begin{bmatrix} R_{RC} \text{ CHILD}_{K} \text{ MAN}_{J} \text{ CL}_{J K} \text{PUSH}_{J} \text{ PE}_{K} \end{bmatrix} \text{ CLICK}$ 'Click on the child that pushes the man.'
 - b. Object relative clause $\begin{bmatrix} & \\ _{RC} \text{ CHILD}_k \text{ CL}_k \text{ MAN}_{j \ j} \text{PUSH}_k \text{ PE}_k \end{bmatrix} \text{ CLICK}$ 'Click on the child that the man pushes.'
 - c. *Filler* [CL TALLEST] CLICK 'Click on the tallest (animal).'



Hauser et al. Glossa: a journal of general linguistics DOI: 10.5334/gjgl.1454

Figure 12 Illustration of an item in LIS.

5.1.3 LSC

For the LSC experiment, each block contained 25 trials (10 subject RCs, 10 object RCs, 5 fillers) preceded by 2 trainings (one example of each type of relative clause). The full list of items is given in the Appendix. The duration of each block was around 15 minutes and was embedded in a larger testing session (3h30) in which participants undertook other lexical and syntactic tests.

Participants were left alone in the experiment room to perform the test on their own. They were sitting approximately 45 cm away from the screen (4:3 display, 22" screen) and answered using a mouse by clicking on the selected character.

For each trial, participants first saw the stimulus video (without context), then clicked on an arrow to transition to the answer picture where they could click on the correct character (see *Figure 13*).

To avoid that participants might interpret any spatial information included in the items topographically, i.e. as referring to the actual position of the characters in the picture rather than a grammatical representation arguments' reference and agreement relation, the picture was only made visible after the end of the recorded sentence. The stimulus video consisted in a request embedding a subject RC (30a) or an object RC (30b). Fillers were either subject RCs and object RCs targeting the middle character or simpler sentences without any embedding like (30c).

(30) *LSC*

- a. Subject relative clause $\begin{bmatrix} RC \\ RC \end{bmatrix}$ SAME **ELEPHANT**_i SPLASH_k BOY_k IX₂ CLICK 'Click on the elephant that splashes the boy.'
- b. Object relative clause $\begin{bmatrix} \\ RC \end{bmatrix} SAME ELEPHANT_i BOY_k SPLASH_i] IX_2 CLICK$ 'Click on the elephant that the boy splashes.'
- c. *Filler* ANIMAL BLUE IX₂ CLICK 'Click on the blue animal.'



5.2 Participants

Across all three sign languages, participants were mainly selected based on their hearing status (deafness had to be attested before 3 years of age) and the age at which they were first exposed to a sign language (no later than 15 y.o.). They were recruited either through social media (in France), through various associations for the Deaf in various cities (in Italy) or within the local Deaf community (Catalonia). They received a compensation of 50 euros (France, Catalonia) or

Hauser et al. Glossa: a journal of general linguistics DOI: 10.5334/gjgl.1454

Figure 13 Illustration of a filler in LSC.

150 euros (Italy) for their participation in the complete battery of tests.¹² We collected metadata regarding their signing background through a questionnaire in sign language (AoE, existence of deaf relatives, modality of communication with parents, siblings, other relatives, type of education whether oralist, bimodal or mostly SL, etc...). The metadata were anonymized.

Additionally, participants had to perform a short, timed, Odd One Out Cognitive Task¹³ to ensure that they were not presenting potential cognitive issues that would affect their performance. This test was made of 2 training items followed by 28 items in which four simultaneous tokens were presented, one of which did not fit the general pattern obeyed by the three others (i.e. an intruder, see *Figure 14*).



Among the participants matching our inclusion criteria (i.e. hearing status, AoE and absence of detected cognitive impairment), we created three "language" groups, based on the age of first sign language exposure, what we call their language group:

- (i) participants with deaf parent(s), exposed to a SL from birth (natives);
- (ii) participants exposed before primary school (early learners, with an AoE ranging between 1 and 5 y.o.) and
- (iii) participants exposed later, between 6 to 15 y.o. (late learners).

The distribution of signers across the different age groups slightly varies across the three sign languages. They are detailed below.

5.2.1 LSF

We collected data from 49 participants (40 in Paris and 9 in Nantes), out of which we excluded seven. One participant was excluded because they were not French, but Belgian, hence signing French-Belgian SL (LSFB); five were excluded due to their very late AoE to a sign language (after 15 y.o.), and one participant was excluded based on their low performance on fillers, with an accuracy below the 75% threshold. Among the 42 remaining participants, 13 were native LSF signers, 14 were early LSF signers and 15 were late LSF signers. Participants were aged between 19 and 72 years (mean = 36.95 y.o., SD = 9.31) and there were 23 women and 19 men.

5.2.2 LIS

We collected data from 44 participants, out of which we excluded five. One participant was excluded due to their very late AoE to a sign language (after 15 y.o.); one participant was excluded because they were postlingually deaf; one because of a poor performance in the

Hauser et al. Glossa: a journal of general linguistics DOI: 10.5334/gjgl.1454

Figure 14 Example of an item within the Odd-one out test.

¹² The battery of tests was longer in Italy, hence the higher compensation.

¹³ The cognitive test administered for the three language groups was the same, but is made accessible in the SIGN-HUB assessment tools platform as three different tests: COGONEOUTLSF (Aristodemo &Friedmann 2019) for LSF; COGONEOUTLIS (Giustolisi & Friedmann 2019) for LIS; COGONEOUTLSC (Zorzi, Sánchez Amat & Friedmann 2019) for LSC.

general cognitive task and two participants were excluded based on their low performance on fillers, with an accuracy below the 75% threshold. Among the 39 remaining participants (19 women, 20 men), 15 were native LIS signers, 13 were early LIS signers and 11 were late LIS signers. Participants were aged between 30 and 65 years (mean = 46.45 y.o., SD = 8.68).

5.2.3 LSC

We collected data from 42 participants at Pompeu Fabra University, out of which we excluded 8. One participant was excluded due to a low score in the cognitive test; one for technical problems in collecting their metadata; one due to their late AoE to LSC (after 15 y.o.), and five participants were not considered based on their low performance on fillers, with an accuracy below the 75% threshold. Among the 34 remaining participants (19 women, 15 men), 12 were native LSC signers, 11 were early LSC signers and 11 were late LSC signers. Participants were aged between 23 and 69 years (mean = 47.67 y.o., SD = 10.6).

6 Results and analyses

6.1 Accuracy

The analyses were realized on each sign language independently, but following the same scripts.¹⁴ We had a binary dataset in which correct answers were coded as "1" and incorrect answers as "0". We started by conducting an item analysis to detect problematic items on which native signers performed below 50% on average. If the score obtained was due to a technical error, we removed the item in both blocks (the subject RC and object RC version of it). If no technical errors could explain the result, the items were kept. In LSF, this led to the removal of four items, hence leaving 48 items (24 subject RC and 24 object RC) to analyze. In LIS, three items were removed, hence leaving 34 items (17 subject RC and 17 object RC) within the analysis. In LSC, only one item had a technical issue such that we performed the analysis on the remaining 38 items (19 subject RC and 19 object RC).

Accuracy and error results were analyzed with the R software (R Core Team 2020) using generalized linear mixed models from the binomial family via the glmer function (package lme4, Bates et al. 2015). The independent variables were language group (native, early & late), condition (subject RC & object RC) and chronological age (as a continuous, mean centered corrected variable). Random variables were intercepts for participants and items.¹⁵

Using ANOVA, we started by comparing minimally different models (df = 1) in which one of the independent variables was removed to control for the overall impact of a given parameter in the data distribution. Keeping only relevant independent variables (more details within each language subsection), we then conducted pairwise comparisons using the package emmeans (Lenth 2020), to obtain finer grained information regarding differences between natives, early, and late learners in their understanding of subject RCs on one side, and of object RCs on the other side.

The results concerning accuracy are presented and analyzed separately for each language here after, error results are presented in Section 5.2.

6.1.1 LSF

Accuracy in LSF is illustrated in *Figure 15*.

In LSF, models including chronological age as a variable did not provide a better fit of the data, so we removed this parameter from the independent variables of the model ($\chi 2(1) = 3.13$, p = 0.07699). Language group ($\chi 2(2) = 8.06$, p = 0.01775), condition ($\chi 2(1) = 346.16$, p < 0.0001) and the interaction between these two factors ($\chi 2(2) = 8.96$, p = 0.01134) were significant. All pairwise comparisons using the resulting generalized mixed model across language group (natives, early, late) and condition (SRC, ORC) are presented in *Table 1*.

¹⁴ Available for consultation here: $https://osf.io/5bdu2/?view_only = 683792bb06d2410facb49037a02a85b5$.

¹⁵ Example of the maximal model formula: $glmer(Value \sim condition*Group*Age_c + (1|ID.participant) + (1|Item), data = Dataset, family=binomial).$



Each line in *Table 1* presents the estimate, standard error, z-ratio and p-value associated with the use of the model to compare the two datasets referred to in the corresponding row. For example, the first row displays the value of the parameters obtained when comparing the score realized by native signers in subject relative clauses accuracy to their performance in object relative clauses.

Comparison						
By condition	Est.	SE	z	р		
Native SRC vs. Native ORC	-3.374	0.544	-6.204	<.0001		
Early SRC vs. Early ORC	-2.410	0.286	-8.419	<.0001		
Late SRC vs. Late ORC	-3.659	0.337	-10.861	<.0001		
Population Natives vs. Early	Est.	SE	z	р		
Native SRC vs. Early SRC	1.891	0.766	2.467	0.0136		
Native ORC vs. Early ORC	0.927	0.557	1.664	0.0961		
Population Natives vs. Late	Est.	SE	Z	р		
Native SRC vs. Late SRC	1.433	0.790	1.814	0.0697		
Native ORC vs. Late ORC	1.719	0.563	3.053	0.0023		
Population Early vs. Late	Est.	SE	z	р		
Early SRC vs. Late SRC	-0.458	0.631	-0.725	0.4682		
Early ORC vs. Late ORC	0.792	0.532	1.489	0.1365		

Table 1 Pairwise comparison of accuracy across language group (native, early, late) and conditions (SRC, ORC), using generalized mixed model and the binomial family in LSF.

To help the reader better understand how to interpret all these values, significant differences are in **bold** font.

The table shows that for all three groups the difference between subject RCs and object RCs was significant, such that subject RCs were better understood. In the comprehension of object RCs, natives performed significantly better than late learners. For completeness sake: native signers outperformed early signers in understanding subject RCs.

Hauser et al. Glossa: a journal of general linguistics DOI: 10.5334/gjgl.1454

Figure 15 Accuracy in object (blue) and subject relative clause (orange) comprehension in LSF, for native (left, n = 13), early (middle, n = 14) and late (right, n =15) learners. Black crosses represent the mean, and black horizontal lines represent the median of each group. The whiskers extend to the nearest data point that is no more than 1.5 times the Inter Quartile Range from the hinges.

6.1.2 LIS

Accuracy in LIS is shown in *Figure 16*.



In LIS, the presence/absence of the factor of chronological age did not impact the general fit of the data ($\chi 2(1) = 1.35$, p = 0.245). On the other hand, language group ($\chi 2(2) = 7.18$, p < 0.001) and its interaction ($\chi 2(2) = 14.6$, p < 0.0001) with condition, significant in itself ($\chi 2(1) = 3.30$, p < 0.01), were both significant. LIS presents more widespread distributions, which do not allow for the identification of a general Subject Advantage in this language. All pairwise comparisons are presented in *Table 2*. Significant differences are in bold font.

Comparison						
By condition	Est.	SE	z	р		
Native SRC vs. Native ORC	0.321	0.226	1.422	0.1551		
Early SRC vs. Early ORC	-0.106	0.205	-0.516	0.6061		
Late SRC vs. Late ORC	-0.864	0.220	-3.920	0.0001		
Population Natives vs. Early	Est.	SE	z	р		
Native SRC vs. Early SRC	0.995	0.430	2.313	0.0207		
Native ORC vs. Early ORC	1.422	0.434	3.276	0.0011		
Population Natives vs. Late	Est.	SE	z	р		
Native SRC vs. Late SRC	0.420	0.441	0.951	0.3414		
Native ORC vs. Late ORC	1.605	0.444	3.615	0.0003		
Population Early vs. Late	Est.	SE	z	р		
Early SRC vs. Late SRC	-0.575	0.447	-1.288	0.1979		
Early ORC vs. Late ORC	0.183	0.445	0.411	0.6808		

Hauser et al. Glossa: a journal of general linguistics DOI: 10.5334/gjgl.1454

Figure 16 Accuracy in object (blue) and subject relative clause (orange) comprehension in LIS (n= 39), for native (left, n = 15), early (middle, n = 13) and late (right, n = 11) learners. Black crosses represent the mean, and black horizontal lines represent the median of each group. The whiskers extend to the nearest data point that is no more than 1.5 times the Inter Quartile Range from the hinges.

Table 2 Pairwise comparison of accuracy across language group (native, early, late) and conditions (SRC, ORC), using generalized mixed model and the binomial family in LIS.

In LIS, we observe that natives significantly outperformed early learners in subject RCs, and that they outperformed both early and late learners in object RCs. The difference between early and late learners was not significant. The only population in which we observed a Subject Advantage is late learners, where we see a significant difference between subject RCs and object RCs.

6.1.3 LSC

In Catalan Sign Language yet another pattern emerged, as can be seen in Figure 17.



Hauser et al. Glossa: a journal of general linguistics DOI: 10.5334/gjgl.1454

Figure 17 Accuracy in object (blue) and subject relative clause (orange) comprehension in LSC, for native (left, n = 12), early (middle, n = 11) and late (right, n = 11) learners. Black crosses represent the mean, and black horizontal lines represent the median of each group. The whiskers extend to the nearest data point that is no more than 1.5 times the Inter Quartile Range from the hinges.

In LSC, the factors of chronological age ($\chi 2(1) = 9.23$, p < 0.001) and its interaction ($\chi 2(1) = 27.9$, p < 0.0001) with condition were both significant. Similarly, condition ($\chi 2(1) = 388.61$, p < 0.0001) and its interaction ($\chi 2(2) = 44.46$, p < 0.0001) with language group (which, taken alone, is not predictive of the data distribution, $\chi 2(2) = 0.75$, p = 0.6861), were significant, as can be seen very clearly in the graph in *Figure 17*. This language seems to present a strong Subject Advantage that increases with age of exposure to LSC. All pairwise comparisons highlighting the language group by condition interaction are presented in *Table 3*. Significant differences are in bold font.

Comparison **By condition** Est. SE z р Native SRC vs. Native ORC -1.605 0.250 -6.433 <.0001 Early SRC vs. Early ORC -2.736 0.267 -10.236 <.0001 Late SRC vs. Late ORC -4.316 0.361 -11.954 <.0001 **Population Natives vs. Early** Est. SE z р Native SRC vs. Early SRC -0.368 0.521 -0.707 0.4795 Native ORC vs. Early ORC 0.762 0.502 1.520 0.1285 **Population Natives vs. Late** Est. SE z р Native SRC vs. Late SRC -1.003 0.554 -1.810 0.0703 Native ORC vs. Late ORC 1.708 3.088 0.0020 0.553 **Population Early vs. Late** SE Est. z р Early SRC vs. Late SRC -0.634 0.520 -1.219 0.2229 Early ORC vs. Late ORC 0.946 0.520 0.0691 1.818

Table 3 Pairwise comparison of accuracy across language group (native, early, late) and conditions (SRC, ORC), using generalized mixed model and the binomial family in LSC.

In LSC, the Subject Advantage was significant across all three groups. When it comes to object RCs, late learners had a significantly lower performance than native learners. The difference

between late and early learners only approaches significance, while there is no significant difference between natives and early learners.

Regarding chronological age, it does not interact with language group but it does impact the performance on conditions, as can be seen in the effect plot in *Figure 18*.¹⁶ The plot shows how the score is affected by the Age variable, which has been scaled and centered, in the object relative clause condition (left panel) and in the subject relative clause condition (right panel). We see that the impact of Age is greater in object RCs than in subject RCs, since the performance is sharply decreasing as age increases.



Hauser et al. Glossa: a journal of general linguistics DOI: 10.5334/gjgl.1454

Figure 18 Plot of the effect of the interaction between Age and Condition on the accuracy within ORC (left) and SRC (right) in LSC. On the x axis, Age has been scaled and centered.

6.1.4 Summary of accuracy results

As for externally headed RCs, we found a clear Subject Advantage in all three LSF populations. This Subject Advantage interacted with language group; with a greater difference in non-native compared to native signers in the comprehension of subject RCs vs. object RCs.

As for internally headed RCs, results were more difficult to summarize. In LIS only late learners displayed a Subject Advantage, while no bias towards either subject RCs or object RCs surfaced in the performance of natives and early learners. In LSC, non-native signers (early and late learners) were below chance¹⁷ when it came to object RCs, suggesting that they systematically interpreted them as subject RCs. We shall discuss below (§6.4) what the implications of these

17 In this study, since there were three characters in the pictures, the chance level is around 30%.

¹⁶ We used the library(effects) and the function "plot(allEffects())" (Fox & Weisberg 2019).

results might be for the grammar(s) of LSC. The mean accuracy in subject RCs tended to be slightly higher in early and native signers, although the difference was not significant. The two data sets concorded in suggesting a strong subject bias in this language.

Before discussing the relevance of these results and drawing any general conclusions, however, it is essential to also conduct an analysis on the type of errors that were made by the participants across groups and conditions.

6.2 Error analysis

Remember that internally headed RCs involve a further complexity that is independent from the subject RCs vs object RCs contrast, that of determining the head which is not explicitly marked through a fixed position. The design of our experiment allows us to tease apart these two difficulties. Participants had the choice between three characters. Hence there were two types of possible errors: (1) participants could click on the wrong side character, which meant they correctly identified the head but not its function within the RC (subject RCs vs object RCs); or (2) participants could click on the middle character, which meant they misunderstood which was the head of the RC (e.g interpreting the stimulus 'the giraffe that the cow is licking' as 'the cow that is licking the giraffe').

Since LSF presents externally headed RCs, the head is always identified by its position on the left of the RC. We thus expect participants to make few errors concerning head identification (few clicks on the middle character), and more errors concerning the gap position, i.e. many clicks on the wrong side characters. This latter type of errors should be more frequent in object RCs due to the Subject Advantage.

In LIS and LSC, on the other hand, since the head is internal, we expect participants to have troubles identifying it and hence they should make more errors clicking the middle character, independently from the subject RCs or object RCs condition. If there is also a Subject Advantage, we expect the distribution of errors concerning only the side character to depend on the subject RCs/object RCs condition, with more errors of this kind in object RCs.

All the analyses have been realized on each sign language independently, following the same procedure. We have created a subset of the data only containing the wrong answers in the object RCs condition. We have a binary dataset in which answers targeting the middle character were coded as "1" and answers targeting the left/right character were coded as "0". This was our dependent variable.

The independent variables were language group (native, early & late), condition (SRC & ORC), and chronological age. We again controlled for relevant parameters and their interactions by conducting ANOVA analysis across minimally different models. Random variables were intercepts for participants and items, except for LSF whose model only had "items" as a random effect (instead of both "items" and "participants"). The great disparity across participants' error figures, with some presenting only one datapoint in SRC, did not leave enough data against which to test for an effect of the participant variable. The results concerning error patterns in the two conditions are discussed separately for each language.

6.2.1 LSF

In LSF, models including chronological age as a variable did not provide a better fit of the data, so we removed this parameter from the independent variables of the model ($\chi 2(1) = 2.24$, p = 0.1341). The effect of condition approached significance ($\chi 2(1) = 2.97$, p = 0.0846), language group had an impact on the data distribution ($\chi 2(2) = 25.05$, p < 0.0001) and the interaction between the two parameters was not significant ($\chi 2(2) = 3.362$, p = 0.1862). We see in *Figure 19* that the type of errors made in subject RCs are equally distributed between the head noun and the position of the gap for all three populations. In object RCs however, a different pattern seems to emerge for late signers, who present a larger proportion of mistakes targeting the side character than those targeting the middle character. This difference in pattern is significantly different from native signers (Est. = -1.37, SE = 0.34, p < 0.0001) and from early signers (Est. = -1.08, SE = 0.27, p < 0.0001), while the latter two do not differ between them (Est. = 0.29, SE = 0.33, p = 0.3779).

24



Hauser et al. Glossa: a journal of general linguistics DOI: 10.5334/gjgl.1454

Figure 19 Proportion of clicks on the competing head noun (middle character), and on the competing gap (left/right characters) in errors for object RC (left panel) and subject RC (right) in LSF. Comparison between natives (left), early (middle) and late (right).

6.2.2 LIS

In LIS, the factor of chronological age did not provide a better fit when integrated in the model $(\chi 2(1) = 0.66, p = 0.4164)$ and was thus removed. Condition $(\chi 2(1) = 0.15, p = 0.6966)$ and language group $(\chi 2(2) = 1.98, p = 0.3724)$, as well as their interaction $(\chi 2(2) = 1.82, p = 0.4025)$ did not had any impact either, confirming what was already visible graphically in *Figure 20*. As expected, errors target mainly the middle character, the competing head noun (68.,5% of the errors) independently from the conditions or the language groups.



Figure 20 Proportion of clicks on the competing head noun (middle character), and on the competing condition (left/ right) in errors for object RCs (left panel) and subject RCs (right) in LIS. Comparison between natives (left), early (middle) and late (left) learners.

6.2.3 LSC

In LSC, the factor of chronological age did not play a role in the data distribution of error types ($\chi 2(1) = 1.38$, p = 0.2393), similarly, the factor of language group ($\chi 2(2) = 3.28$, p = 0.2194) and its interaction with condition ($\chi 2(2) = 2.15$, p = 0.3407) did not affect the fit. We thus only kept the factor of condition as an independent variable, since it was the only parameter affecting the data distribution significantly ($\chi 2(1) = 26.01$, p < 0.0001). With the resulting model, we obtain a significant difference between subject RCs and object RCs (Est. = 1.46, SE = 0.3, p < 0.0001).

When we look at *Figure 21*, we indeed see that, in subject RCs, all participants (native, early and late) patterned like LIS participants: most errors were of head identification (the middle character represents 78% of errors in subject RCs).



Hauser et al. Glossa: a journal of general linguistics DOI: 10.5334/gjgl.1454

Figure 21 Proportion of clicks on the competing head noun (middle character), and on the competing condition (left/ right) in errors for object RCs (left panel) and subject RCs (right) in LSC. Comparison between natives (left), early (middle) and late (right) learners.

In object RCs, however, a difference between natives on the one side and early learners on the other side arose. Native learners present a pattern that is very similar to what we observed with subject RCs: around 62% or errors were of head identification (middle character). Early learners, on the other hand, present a pattern that is similar to what is observed in LSF, with most errors regarding the identification of the function of the head (60% of errors targeting the side characters). The difference between Native ORC error pattern and Early ORC error pattern is significant (Est. = -1.39, SE = 0.65, p = 0.0341).

7 Discussion

In this paper we had three main research objectives: i) investigate whether the Subject Advantage in the comprehension of RCs extends across modalities; ii) investigate whether a delayed exposure to (sign) language may impact RC comprehension in adult signers; iii) investigate whether the Subject Advantage emerges across relativization strategies, in internally headed RCs as well as in externally headed RCs. In this section we discuss our results relative to these questions and comment on some interesting differences that emerged between LIS and LSC.

7.1 Does the Subject Advantage in RCs comprehension extend across modalities?

Hauser et al. Glossa: a journal of general linguistics DOI: 10.5334/gjgl.1454

The answer to this question is straightforward. We found that externally headed relative clauses present a Subject Advantage across modalities. LSF behaves exactly like other languages that display externally headed relatives, exhibiting a strong Subject Advantage, hence confirming the results reported by Hauser & Pozniak (2019). The Subject Advantage interacts with language group: in LSF, even if it is observable across groups, it is strongest in late learners. In LSC the Subject Advantage is also present in all three groups, and the strength of the effect increases with the age of exposure. In LIS, the Subject Advantage is significant only in the late group.

7.2 Does delayed exposure to (sign) language impact RC comprehension in adult signers?

The answer to this question is also straightforward: the Subject Advantage is modulated by age of exposure. As we summarized in 6.1, the later the exposure, the stronger the Subject Advantage. That difficult structures are more affected by age of exposure is in line with the literature (Emmorey & Corina 1990; Mayberry 1993; Emmorey et al. 1995; Boudreault & Mayberry 2006; Cormier et al. 2012) and with new further evidence by our research group (Authors, under review).

7.3 Does the Subject Advantage emerge across relativization strategies?

As for the third research question concerning internally headed RCs, the pattern of results that we found requires a detailed discussion. As expected, we found that it is more difficult to identify the head in internally headed RCs than in externally headed RCs. In LSC and LIS, more errors were made regarding the identification of the head noun (i.e. participants hesitated between the correct head noun and the middle character), while the opposite pattern was found in LSF, where signers made more mistakes of gap identification, hence targeting the wrong peripheral character.

The result that was most difficult to anticipate (and potentially the most interesting) is the existence of a Subject Advantage in the comprehension of internally headed RCs, which is observable both in LIS and in LSC although to a different degree. In LIS, this Subject Advantage is statistically significant only in late learners, while in LSC it is visible in all three acquisition groups. We think that this result can contribute indirect evidence both on the debate about the source of the Subject Advantage in RCs in general, and for the syntactic analysis to be given to internally headed RCs.

As for the debate about its source, the fact that the Subject Advantage is attested despite the absence of a gap in internally headed RCs is a powerful argument against all approaches that explain the Subject Advantage in languages like English as a reflex of shorter retention of the subject gap in the memory buffer given its vicinity to the relative clause head. Similarly, the presence of a Subject Advantage in internally headed RCs is also a challenge for any explanation that reduces it to a canonicity effect. Swinney, Prather & Love (1996) and Sekerina (2003) suggest that the language processor employs a canonical sentence schema to interpret sentences. Thus, structures in a language's canonical word order (like subject relatives in English) are assumed to be easier to process than structures with a non-canonical order (like object relatives in English or in LSF). Crucially, both subject and object RCs follow the canonical word order in LIS and LSC and, this notwithstanding, a difference emerges between them. Therefore, the Subject Advantage is not likely to be a canonicity effect.

The type of explanation that seems to be best suited to explain the Subject Advantage in internally headed RCs is the account that assumes that (i) there is a processing cost when the subject intervenes hierarchically between the object and its gap (cf. Friedmann et al. 2009 for this explanation of the Subject Advantage) and (ii) the gap that is not visible in overt syntax is created by covert movement of the RC head (cf. Cole, 1987 a.o. for this approach to internally headed RCs).

We acknowledge that with internally headed RCs, namely in LIS and LSC, the Subject Advantage was especially (or only) significant in late learners or non-native populations. Remember that our experiment is an off-line sentence-to-picture matching task and we measured only accuracy, therefore we cannot exclude that a Subject Advantage could emerge even with native LIS and LSC signers with a more sensitive on-line task. Still, it is true that the same off-line task revealed in LSF a Subject Advantage in all groups of signers, including natives. This crosslinguistic difference may suggest that the effect is stronger in externally headed RCs than in internally headed RCs. Notice that this is coherent with the acquisition pattern observed in Korean, where internally headed RCs emerge earlier than externally headed RCs and might perhaps explain the Object Advantage observed in Wenzhounese (Hu Shenai et al. 2018). The modulation of the Subject Advantage might also indicate that the two factors mentioned above (canonicity effects and presence versus absence of an overt gap) might play a secondary role, adding to the main effect due to hierarchical subject intervention in object dependencies.

As we mentioned, the Subject Advantage in internally headed RCs provides indirect evidence in favor of a certain kind of analysis of this structure, one that assumes a covert movement of the head. This has an important consequence: if we are right, the Subject Advantage, an effect observable in comprehension that is easily measurable with experimental methods, can be used as a reliable, if indirect, diagnostic for LF movement. As such, it could be used in addition to islands effects (traditionally employed to argue for a covert movement analysis of wh-in situ at least since Huang Charles 1982) in all those cases where a covert dependency is argued for and needs to be detected (cf. also Xiang Ming et al. 2014; 2015 for experiments aiming at other direct processing evidence for this covert dependency). This is an interesting case of cross-modal interaction in which a diagnostic detected in a sign language can be extended to the analysis of spoken languages having the same abstract configuration.

7.4 Comparing LIS and LSC

LSC and LIS did not yield identical results while both displaying internally headed RCs, and both exhibiting a Subject Advantage. In LIS only late learners display a significant Subject Advantage, while native and early learners appear to have a good comprehension of both subject and object RCs. In LSC, the Subject Advantage is visible across all signers, with a clear distinction between native and non-native comprehension of object relative clauses.

We might wonder here about the source of this difference. There are two possible ways to go. One is related to a methodological difference between the two experiments and the other concerns a grammatical difference between the two languages.

Comparing the design between the two experiments, we observe that, while in the LIS experiment, the sentence was shown together with a miniature of the response picture, this was never the case in LSC, where the sentence was shown before the picture. It could be the case that the simultaneous exposure to the picture and the sentence induce LIS participants to a direct (non-structurally mediated) mapping between position of the sign in the neutral space and position of the corresponding character in the picture. Notice however that this is not likely, because the LIS team controlled the possible mapping between spatial organization of the signs in the sentence and that of the characters in the picture: while in half of the items, the loci associated to the arguments in the sentence indeed matched the positions associated with the characters in the response picture, the other half of the items displayed a systematic mismatch between grammatical space and the disposition of the characters in the picture. Had the participants relied on such a topographical mapping, they would have made many more mistakes in general, and crucially more than in the LSC experiment, where any topographical mapping was discouraged by the non-simultaneous display of the sentence and the picture. This means that the explanation of the difference between LSC and LIS data has to be found elsewhere.

Turning to the linguistic side, the internal make up of internally headed RCs in the two languages is different, in a way that might contribute to explaining the differences we observe in comprehension data.

In LIS, the relative marker PE obligatorily agrees in space with the internal head in LIS, as shown in the object relative clause (23) visually represented in *Figure 6*, repeated below as *Figure 22*.

28



In *Figure 22* the relative PE points towards the locus in space associated with the object classifier co-referring with the giraffe, which is the (topicalized) head of the RC. Participants are likely to rely on this spatial agreement information to interpret the RC.

No such explicit spatial agreement is displayed in LSC relative clauses, where the relative marker MATEIX ('SAME') is articulated on the body. As can be observed in *Figure 9*, repeated here as *Figure 23*, corresponding to the object RC (27), the association between the relative marker SAME and the head (GIRAFFE in the example) is only signaled through a light head tilt and body turn.



On the other hand, in this example the relative element SAME sits close to the (topicalized) head of the RC, and participants are likely to use the linear proximity to identify the head of the RC. This cue is however less reliable than spatial agreement in LIS since the position of SAME is not systematically adjacent to the RC head in LSC grammar (Mosella Sanz 2012), even if this was its position in our data set. We speculate that this difference in the grammar of the two languages, and in particular the absence of a grammatical marker of the head, like spatial agreement, explains why the Subject Advantage is stronger in LSC: in the absence of a univocal cue identifying the head of the RC, participants are more prone to follow their bias towards a subject interpretation.

If we look now closer to LSC, we find that the Subject Advantage is so strong as to take the shape of a categorical difference between the grammar of native signers and that of non-native signers who systematically misunderstand object relative clauses.

The accuracy and error data that we observed are compatible with the hypothesis that native signers have both subject RCs and object RCs in their grammar while non-native signers do not allow object RCs at all in LSC. This configuration offers yet another illustration of the Accessibility Hierarchy and its Primary Relativization Constraint corollary (Keenan & Comrie 1977), which states that:

- 1. A language must have a primary RC-forming strategy.
- **2.** If a primary strategy in a given language can apply to a low position on the Accessibility Hierarchy, then it can apply to all higher positions.
- 3. A primary strategy may cut off at any point on the Accessibility Hierarchy.

Hauser et al. Glossa: a journal of general linguistics DOI: 10.5334/gjgl.1454

Figure 22 Screen captures of the sentence (23), an object relative clause in LIS.

Figure 23 screen capture of the sentence in (27), an object relative clause in LSC.

The LSC situation appears to comply with the Primary Relativization Constraint: (1) is verified since all signers of LSC understand subject RCs; (2) is verified since native signers, who understand object RCs, also understand subject RCs and (3) points towards two different grammars between native and non-native signers: the native grammar cuts off the Accessibility Hierarchy after object RCs, while the non-native grammar cuts it off right after SRCs. The fact that different varieties of the same language realize different steps of the Accessibility Hierarchy should not come as a surprise given the exceptional circumstances of access to language experienced by a large part of the deaf population. In fact, this finding, which replicates language internally the conclusion based on the typological literature, appears as an extreme case of AoE effect.

8 Conclusion

This article sheds new light on the cross-modality and cross-typology biases that affect relative clause comprehension, showing that a) the Subject Advantage described by Keenan & Comrie's Accessibility Hierarchy (1977) holds also for sign languages and b) it extends across relative constructions, affecting in particular an understudied relativization strategy, internally headed relative clauses. Altogether the results of our study confirm the validity of a structural approach to asymmetries in relative clauses comprehension, as well as the importance of the Subject Advantage as a diagnostic for long distance dependencies, be they overt or covert. As for the impact of age of exposure, we confirmed that native signers outperform non-native signers and that delay in language exposure significantly affects adults' language competence. We found that non-native signers exhibit a stronger Subject Advantage in the three sign languages we investigated, and that this takes the form of a categorial grammatical difference in LSC, where the language of non-native signers appear to lack object relativization.

Data Accessibility Statement

All the video items used in the experiments and the scripts for the analyses can be found in the following OSF repository: *https://osf.io/5bdu2/?view_only* = 683792bb06d2410facb49037a02a85b5. Images can be consulted by contacting the corresponding author. The detailed list of items if provided here after.

Abbreviations

ACC = accusative, a.o = among others, AoE = Age of First Exposure; ASL = American Sign Language, bt = body turn, CL = classifier, DGS = German Sign Language (*Deutsche Gebärdensprache*), df = degrees of freedom, EEG = electroencephalogram, fMRI = functional magnetic resonance imaging, GEN = genitive, HKSL = Hong Kong Sign Language (香港手語), ICED = International Congress for the Education of the Deaf, JSL = Japanese Sign Language (日本手話, Nihon Shuwa), LAN = Left Anterior Negativity, Libras = Brazilian Sign Language (*Língua de Sinais Brasileira*), LIS = Italian Sign Language (Lingua dei Segni Italiana), LOC = locative, LSC = Catalan Sign Language (*Llengua de Signes Catalana*), LSF = French Sign Language (*Langue des Signes Française*), LSFB = French Belgian Sign Language (*Langue des Signes Belge Francophone*), NOM = nominative case, OPART = object partitive, ORC = object relative clause, PET = positron emission tomography, PL = plural, RC(s) = relative clause(s), re = raised eyebrows, rel = non-manual markers associated with relative clauses, rg = right gaze, SD = standard deviation, SG = singular, SL = sign language, SOV = subject object verb, SPART = subject partitive, SRC = subject relative clause, SVO = subject verb object, TiD = Turkish Sign Language (*Türk İşaret Dili*), TOP = topic, y.o. = years old.

Additional file

The additional file for this article can be found as follows:

Appendix. Material. DOI: https://doi.org/10.5334/gjgl.1454.s1

30

Ethics and consent

We declare that the authors have no conflict of interest to declare. The three experiments reported meet the requirements of informed consent as laid out in European and National ethical guidelines (see art. 19 of Regulation of Establishment of Horizon 2020 No 1291/2013; CNIL for France, NBC for Italy, CEIC for Spain). They received the previous approval of local ethical committees in France (CERES, IRB n° 20163400001072), Italy (Milan Bicocca Ethical Committee, prot. n° 0019845/16) and Spain (Parc de Salut MAR – Clinical Research Ethics Committee, prot. n° 2016/6715/I). The comprehension tests that were used for the experiments are part of the SIGN-HUB assessment tests and are named respectively SYNTRELLSF- *Relative clauses comprehension task in LSF* (Hauser et al. 2019), SYNTRELLIS- *Relative clauses comprehension task in LSC* (Zorzi et al. 2019). They are all accessible and usable for assessment on the SIGN-HUB platform under request: https://www.sign-hub.eu/assessment/welcome-page-assessment.

Acknowledgements

Many people participated to the discussions, brain storming sessions, and technical struggles which are behind the results discussed here and need to be acknowledged. Within the SIGN-HUB network, Josep Quer and Naama Friedmann gave a crucial contribution, and we are grateful for this.

On the French side, we must thank Marion Bonnet, who helped us with the platform as part of her internship, and Chiara Annucci and Angélique Jaber, who helped with the administration of the tests. Nothing could have been possible without our consultant, Valérie Jendoubi, and all the participants to the experiment. On the Italian side, thanks are due to all participants and to the Instituto dei Sordi of Turin for hosting us. We are grateful to Alessandra Checchetto and Rita Sala for many useful discussions and suggestions and to Cecilia Gazzeri and Luca Falbo who helped with the administration of the tests. Last but not least thanks to our consultants: Antonino D'Urso, Rosella Ottolini, Mauro Mottinelli, Mirko Pasquotto, Filippo Calcagno e Gabriele Caia. On the Catalan side, special thanks are due to all participants and to our consultants Santiago Frigola Segimon and Delfina Aliaga Emeterio, who, together with Francisco Javier Alavarez Reyes, also helped with the administration of the tests. We need to thank the GrEP research group at UPF for letting us use their testing rooms. We need to thank Marianthi Koraka who helped us with the platform as part of her internship at UPF, and we are grateful to Josep Quer for useful discussions and support throughout the whole creation and administration of the tests.

Funding information

The research summarized in this paper is part of the SIGN-HUB project, which has received funding from the European Union's Horizon 2020 research and innovation program under grant agreement No 693349.

Competing interests

The authors have no competing interests to declare.

Author contributions

VA, BG, GZ, JSA, CD and CC contributed to the study conception and design. CD and CC supervised the project and acquired the funding. LSF material, which was used as a baseline for the other sign languages, was prepared by CH, VA and CD and LSF data collection was performed by CH. LIS material preparation and coordination of LIS data collection were performed by BG, RS and CC. LSC material preparation and coordination of LSC data collection were performed by GZ and JSA. Data analysis was performed by CH, BG, and DG. All authors discussed the results. The first draft of the manuscript was written by CH, CD and CC. CH and CD made all the ameliorations suggested by the reviewers. All authors approved the final manuscript.

Author affiliations

Charlotte Hauser (D) orcid.org/0000-0001-5444-3446 Université de Paris - LLF, CNRS. 8 place Paul Ricœur, 75013, Paris Giorgia Zorzi D orcid.org/0000-0003-2427-2732 Universitat Pompeu Fabra, Roc Boronat, 138, 08018, Barcelona Valentina Aristodemo D orcid.org/0000-0003-1667-7772 Université de Paris - LLF, CNRS. 8 place Paul Ricœur, 75013, Paris Beatrice Giustolisi D orcid.org/0000-0001-5967-1392 Università degli Studi di Milano-Bicocca, Piazza dell'Ateneo Nuovo, 1 - 20126, Milano Doriane Gras (D) orcid.org/0000-0002-6198-1450 Université de Paris - LLF, CNRS. 8 place Paul Ricœur, 75013, Paris Rita Sala D orcid.org/0000-0003-2055-8370 Université Paris 8 - SFL, CNRS. 59/61 rue Pouchet, 75849 Paris cedex 17, France Jordina Sánchez Amat D orcid.org/0000-0002-9854-0499 Universitat Pompeu Fabra, Roc Boronat, 138, 08018, Barcelona Carlo Cecchetto D orcid.org/0000-0001-9985-7087 Università degli Studi di Milano-Bicocca, Piazza dell'Ateneo Nuovo, 1 - 20126, Milano; Université Paris 8 - SFL, CNRS. 59/61 rue Pouchet, 75849 Paris cedex 17, France Caterina Donati D orcid.org/0000-0002-7076-3340 Université de Paris - LLF, CNRS. 8 place Paul Ricœur, 75013, Paris

References

- Aoun, E. Joseph, Elabbas Benmamoun & Lina Choueiri. 2010. The Syntax of Arabic. Cambridge: Cambridge University Press. DOI: https://doi.org/10.1017/CB09780511691775
- Barberà, Gemma. 2020. Syntax: 3.4. Relative Clauses. In Josep Quer & Gemma Barberà (eds.), A Grammar of Catalan Sign Language (LSC). 1st ed. (SIGN-HUB Sign Language Grammar Series), 230–237. http:// sign-hub.eu/grammars/...)
- Bates, Douglas, Martin Maechler, Ben Bolker & Steve Walker. 2015. Fitting Linear Mixed-Effects Models using lme4. *Journal of Statistical Software* 67(1). 1–48. DOI: https://doi.org/10.18637/jss.v067.i01
- Benedicto, Elena & Diane Brentari. 2004. Where did all the arguments go?: Argument-changing properties of classifiers in ASL. *Natural Language & Linguistic Theory* 22(4). 743–810. DOI: https://doi. org/10.1007/s11049-003-4698-2
- Bianchi, Valentina. 2000. The raising analysis of relative clauses: A reply to Borsley. *Linguistic Inquiry* 31(1). 123–140. DOI: <u>https://doi.org/10.1162/002438900554316</u>
- Boudreault, Patrick & Rachel I. Mayberry. 2006. Grammatical processing in American Sign Language: Age of first-language acquisition effects in relation to syntactic structure. *Language and cognitive processes* 21(5). 608–635. DOI: *https://doi.org/10.1080/01690960500139363*
- Branchini, Chiara. 2014. On relativization and clefting: An analysis of Italian Sign Language. The Hague: de Gruyter. DOI: https://doi.org/10.1515/9781501500008
- Branchini, Chiara & Caterina Donati. 2009. Relatively different: Italian Sign Language relative clauses. In Anikó Liptak (rd.), *Correlatives cross-linguistically*, 157–191. Amsterdam: Benjamins. DOI: https://doi. org/10.1075/lfab.1.07bra
- Cecchetto, Carlo, Carlo Geraci & Alessandro Zucchi. 2006. Strategies of relativization in Italian Sign Language. *Natural Language and Linguist Theory* 24(4). 945–975. DOI: *https://doi.org/10.1007/ s11049-006-9001-x*
- Cecchetto, Carlo & Caterina Donati. 2015. (Re) Labeling. Cambridge MA: The MIT Press. DOI: https://doi. org/10.7551/mitpress/9780262028721.001.0001
- Chen, Jidong & Yasuhiro Shirai. 2015. The acquisition of relative clauses in spontaneous child speech in Mandarin Chinese. *Journal of child language* 42(2). 394–422. DOI: *https://doi.org/10.1017/S0305000914000051*
- Cinque, Gugliemo. 2019. The different merge positions of the different types of relative clauses. In Ken Ramshøj Christensen, Henrik Jørgensen & Johanna Wood (eds.), *The Sign of the V – Papers in Honour of Sten Vikner*, 131–147. Dept. of English, School of Communication & Culture, Aarhus University. DOI: https://doi.org/10.7146/aul.348.93
- Cole, Peter. 1987. The structure of internally headed relative clauses. *Natural Language & Linguistic Theory* 5(2). 277–302. DOI: https://doi.org/10.1007/BF00166587
- Constable, Robert T., Kenneth R. Pugh, Ella Berroya, W. Eynar Mencl, Michael Westerveld, Weijia Ni & Donald Shankweile. 2004. Sentence complexity and input modality effects in sentence comprehension: an FMRI study. *Neuroimage* 22. 11–21. DOI: *https://doi.org/10.1016/j. neuroimage.2004.01.001*

- Cormier, Kearsy, Adam Schembri, David Vinson & Eleni Orfanidou. 2012. First language acquisition differs from second language acquisition in prelingually deaf signers: Evidence from sensitivity to grammaticality judgement in British Sign Language. *Cognition* 124(1). 50–65. DOI: https://doi. org/10.1016/j.cognition.2012.04.003
- Culy, Chirstopher D. 1991. The syntax and semantics of internally headed relative clauses. Stanford: Doctoral dissertation.
- De Vries, Mark. 2001. Patterns of relative clauses. *Linguistics in the Netherlands* 8. 231–243. DOI: https:// doi.org/10.1075/avt.18.23vri
- DeCasper, Anthony J., & Melanie J. Spence. 1986. Prenatal maternal speech influences newborns' perception of speech sounds. *Infant behavior and Development* 9(2). 133–150. DOI: *https://doi.org/10.1016/0163-6383(86)90025-1*
- Diessel, Holger. 2004. The acquisition of complex sentences. Cambridge; Cambridge University Press. DOI: https://doi.org/10.1017/CBO9780511486531
- Diessel, Holger & Michael Tomasello. 2005. A new look at the acquisition of relative clauses. *Language* 81(4). 882–906. DOI: *https://doi.org/10.1353/lan.2005.0169*
- Emmorey, Karen & David Corina. 1990. Lexical recognition in sign language: Effects of phonetic structure and morphology. Perceptual and motor skills 71, 1227–1252. DOI: https://doi.org/10.2466/pms.1990.71.3f.1227
- Emmorey, Karen, Ursula Bellugi, Angela Friederici & Petra Horn, P. 1995. Effects of age of acquisition on grammatical sensitivity: Evidence. *Applied Psycholinguistics* 16(1). 1–23. DOI: https://doi.org/10.1017/ S0142716400006391
- Fontana, Joseph M. 1990. Is ASL like Diegueño or Diegueño like ASL? A study of internally headed relative clauses in ASL. In Ceil Lucas (ed.), *Sign language research. Theoretical issues*, 238–255. Washington, DC: Gallaudet University Press.
- Fox, John, & Sanford Weisberg. 2019. An R Companion to Applied Regression, 3rd Edition. Thousand Oaks, CA < https://socialsciences.mcmaster.ca/jfox/Books/Companion/index.html>.
- Frazier, Lynn. 1987. Sentence processing: A tutorial review. In M. Coltheart (ed.). *Attention and performance 12: The psychology of reading*, 559–586. Lawrence Erlbaum Associates.
- Friedmann, Naama, Adriana Belletti & Luigi Rizzi. 2009. Relativized relatives: Types of intervention in the acquisition of A-bar dependencies. *Lingua* 119(1). 67–88. DOI: https://doi.org/10.1016/j. lingua.2008.09.002
- Friedmann, Naama & Rama Novogrodsky. 2004. The acquisition of relative clause comprehension in Hebrew: A study of SLI and normal development. *Journal of Child language* 31(3). 661–681. DOI: https://doi.org/10.1017/S0305000904006269
- Galloway, Teresa. 2011. Why agreement matters: The syntax of relative clauses in ASL. In Poster presented at the conference Formal and Experimental Approaches to Sign Language Theory (FEAST), Venice.
- Gervain, Judit, & Janet F. Werker. 2008. How infant speech perception contributes to language acquisition. *Language and Linguistics Compass* 2(6). 1149–1170. DOI: *https://doi.org/10.1111/j.1749-818X.2008.00089.x*
- Gibson, Edward. 2000. The dependency locality theory: A distance-based theory of linguistic complexity. *Image, Language, Brain* 2000, 95–126.
- Hale, John. 2016. Information-theoretical complexity metrics. Language and Linguistics Compass 10. 397– 412. DOI: https://doi.org/10.1111/lnc3.12196
- Hall, Matthew L., Wyatte C. Hall & Naomi K. Caselli. 2019. Deaf children need language, not (just) speech. *First Language* 39(4). 367–395. DOI: https://doi.org/10.1177/0142723719834102
- Happ, Daniela & Marc-Oliver Vorköper. 2006. *Deutsche Gebärdensprache. Ein Lehr- und Arbeitsbuch*. Frankfurt: Fachochschulverlag.
- Haug, Tobias. 2005. Review of sign language assessment instruments. Sign Language & Linguistics 8(1–2). 61–98. DOI: https://doi.org/10.1075/sll.8.1.04hau
- Hauser, Charlotte. 2016. Relative clauses in LSF. *Typology and analysis*. Master's thesis Ecole des hautes études en sciences sociales.
- Hauser, Charlotte. 2019. Subordination in LSF: nominal and sentential embedding. Université de Paris Doctoral dissertation.
- Hauser, Charlotte & Carlo Geraci. 2017. Relativization Strategies in LSF, some preliminary results. *Proceedings of the FEAST 2017 conference*, 16–26.
- Hauser, Charlotte & Céline Pozniak. 2019. Relative clauses processing in LSF: an eye-tracking innovative experiment. *Poster presented at at AMLAP 2019th* Moscow.
- Hawkins, John A. 1999. Processing complexity and filler-gap dependencies across grammars. Language 75(2). 244–285. DOI: https://doi.org/10.2307/417261
- Hu, Shenai, Carlo Cecchetto & M. Teresa Guasti. 2018. A new case for structural intervention: evidence from Wenzhounese relative clauses. *Journal of East Asian Linguistics* 27(3). 247–273. DOI: https://doi. org/10.1007/s10831-018-9182-4
- Huang, Charles. T. J. 1982. Move wh in a language without wh-movement. *The linguistic review* 1(4). 369–416. DOI: *https://doi.org/10.1515/tlir.1982.1.4.369*

- Humphries, Tom, Poorna Kushalnaga, Gaurav Mathur, Donna Jo Napoli, Carol Padden, Christian Rathmann, & Scott Smith. 2016. Language choices for deaf infants: Advice for parents regarding sign languages. *Clinical Pediatrics* 55(6). 513–517. DOI: https://doi.org/10.1177/0009922815616891
- Jäger, Lena, Zhong Chen, Qiang Li, Chien-Jer Charles Lin & Shravan Vasishth. 2015. The subject-relative advantage in Chinese: Evidence for expectation-based processing. *Journal of Memory and Language* 79. 97–120. DOI: https://doi.org/10.1016/j.jml.2014.10.005
- Jeon, K. Seon & Hae-Young Kim. 2007. Development of relativization in Korean as a foreign language: The noun phrase accessibility hierarchy in head-internal and head-external relative clauses. *Studies in Second Language Acquisition* 29(2). 253–276.
- Kayne, Richard S. 1994. The antisymmetry of syntax. Camnbridge, MA: The MIT press.
- Keenan, Edward L. & Bernard Comrie. 1977. Noun phrase accessibility and universal grammar. *Linguistic inquiry* 8(1). 63–99.
- Keenan, Edward L. & Sarah Hawkins. 1987. The psychological validity of the Accessibility Hierarchy. In Edward L. Keenan (eds.). *Universal grammar: 15 essays*, 60–85. London: Croom Helm.
- Kim, Young-Joo. 1987. The acquisition of relative clauses in English and Korean: Development in spontaneous production. Hanshin Publishing Company.
- King, Jonathan & Marcel Adam Just. 1991. Individual differences in syntactic processing: The role of working memory. *Journal of memory and language* 30(5). 580–602. DOI: https://doi. org/10.1016/0749-596X(91)90027-H
- Kornfilt, Jaklin. 1997. On the syntax and morphology of relative clauses in Turkish. *Dilbilim Araştırmaları Dergisi 8*, 24–51.
- Kubuş, Okan. 2011. Relative clause constructions in Turkish Sign Language (Turk Isaret Dili -TİD). Paper presented at the Nijmegen Gesture Centre lecture series.
- Kwon, Nayoung, Yoonhyoung Lee, Peter C. Gordon, Robert Kluender & Maria Polinsky. 2010. Cognitive and linguistic factors affecting subject/object asymmetry: An eye-tracking study of prenominal relative clauses in Korean. *Language* 86(3). 546–582. DOI: <u>https://doi.org/10.1353/ lan.2010.0006</u>
- Lane, Harlan. 1979. Histoire chronologique de la répression de la langue des signes en France et aux Etats-Unis. *Langages* 56. 92–124. DOI: *https://doi.org/10.3406/lgge.1979.1831*
- Lau, Elaine. 2016. Acquisition of relative clauses in Cantonese: A multi-factorial analysis University of Hawai'i at Manoa Doctoral Dissertation.
- Lee, Kwee Ock. 1991. On the first language acquisition of relative clauses in Korean: The universal structure of COMP. Cornell University Doctoral dissertation.
- Lehmann, Christian. 1986. On the typology of relative clauses. *Linguistics* 24(4). 663–680. DOI: https:// doi.org/10.1515/ling.1986.24.4.663
- Lenth, Russell. 2020. emmeans: Estimated Marginal Means, aka Least-Squares Means. R package version 1.5.2-1. *https://CRAN.R-project.org/package = emmeans*.
- Li, J. 2013. Relative constructions in Hong Kong Sign Language. The Chinese University of Hong Kong Doctoral Dissertation.
- Liddell, Scott K. 1978. Nonmanual signals and relative clauses in American Sign Language. In P. Siple (ed.). *Understanding language through sign language research*, 59–90. New York, NY: Academic Press.
- Liddell, Scott K. 1980. American Sign Language syntax. The Hague: Mouton. DOI: https://doi. org/10.1515/9783112418260
- Lin, Chien-Jer Charles & Thomas G. Bever. 2006. Subject preference in the processing of relative clauses in Chinese. In *Proceedings of the 25th west coast conference on formal linguistics* 25. 254–260. Somerville, MA: Cascadilla.
- Mak, Willem M., Wietske Vonk & Herbert Schriefers. 2006. Animacy in processing relative clauses: The hikers that rocks crush. *Journal of memory and language* 54(4). 466–490. DOI: *https://doi.org/10.1016/j.jml.2006.01.001*
- Mayberry, Rachel I. 1993. First-language acquisition after childhood differs from second-language acquisition: The case of American Sign Language. Journal of Speech, Language, and Hearing Research 36(6). 1258–1270. DOI: https://doi.org/10.1044/jshr.3606.1258
- Mecklinger, Axel, Herbert Schriefers, Karsten Steinhauer & Angela D. Friederici. 1995. Processing relative clauses varying on syntactic and semantic dimensions: An analysis with event-related potentials. *Memory & Cognition* 23(4). 477–494. DOI: https://doi.org/10.3758/BF03197249
- Miller, Dana. 1990. ASL relative clauses and their interaction with null categories. In Ceil Lucas (ed.), Sign language research. Theoretical issues, 223–237. Washington, DC: Gallaudet University Press.
- Miyamoto, Edson T. & Michiko Nakamura. 2013. Unmet expectations in the comprehension of relative clauses in Japanese. In *Proceedings of the Annual Meeting of the Cognitive Science Society* 35. 3074–3079.

- Mosella Sanz, Marta. 2012. Les construccions relatives en llengua de signes catalana (LSC) University of Barcelona Doctoral Dissertation.
- Müller, Horst M., Jonathan W. King & Marta Kutas 1997. Event-related potentials elicited by spoken relative clauses. *Cognitive Brain Research* 5. 193–203. DOI: https://doi.org/10.1016/S0926-6410(96)00070-5
- Nunes, Jairo & Ronice Müller de Quadros. 2005. Duplication of wh-elements in Brazilian Sign Language. In Proceedings of the 35th Conference of the North Eastern Linguistic Society.
- O'Grady, William, Miseon Lee & Miho Choo 2003. A subject-object asymmetry in the acquisition of relative clauses in Korean as a second language. *Studies in Second Language Acquisition* 25. 433–448. DOI: https://doi.org/10.1017/S0272263103000172
- Özge, Duygu, Theodoros Marinis & DenizZeyrek. 2009. Comprehension of subject and object relative clauses in monolingual Turkish children. In *14th International Conference on Turkish Linguistics*, 341–350.
- Padden, Carol. 1988. Interaction of morphology and syntax in American Sign Language. New York: Garland.
- Penner, Mark, Uiko Yano & Hiyeda Terasawa 2019. Relative Clauses in Japanese Sign Language. *Minpaku Sign Language Studies* 2. 1–24.
- Pfau, Roland & Markus Steinbach. 2005. Relative clauses in German Sign Language: extraposition and reconstruction. In Leah Bateman & Cherlon Ussery (eds.). *Proceedings of NELS 35*, 507–521. GLSA Amherst.
- Qiao, Xiaomei, Liyao Shen & Kenneth Forster. 2012. Relative clause processing in Mandarin: Evidence from the maze task. *Language and Cognitive Processes* 27(4). 611–630. DOI: *https://doi.org/10.1080/0* 1690965.2011.578394
- R Core Team (2020). R: A language and environment for statistical computing. Vienna, Austria: R Foundation for Statistical Computing. URL *https://www.R-project.org/*.
- Roland, Dougals, Gail Mauner, Carolyn O'Meara & Hongoak Yun. 2012. Discourse expectations and relative clause processing. *Journal of Memory and Language* 66(3). 479–508. DOI: https://doi.org/10.1016/j.jml.2011.12.004
- Sauerland, Uli. 2003. Unpronounced heads in relative clauses. *The interfaces: Deriving and interpreting omitted structures* 61. 205–226. DOI: https://doi.org/10.1075/la.61.10sau
- Sekerina, Irina A. 2003. Scrambling and processing: dependencies, complexity, and constraints. In S. Karimi (ed.). Word order and scrambling, 301–324. Oxford: Blackwell. DOI: https://doi. org/10.1002/9780470758403.ch13
- Shimoyama, Junko. 1999. Internally headed relative clauses in Japanese and E-type anaphora. *Journal of East Asian Linguistics* 8(2). 147–182. DOI: https://doi.org/10.1023/A:1008338020411
- Slobin, Dan I. & Karl Zimmer (Eds.). 1986. Studies in Turkish linguistics. Amsterdam: Benjamins. DOI: https://doi.org/10.1075/tsl.8
- Stromswold, Karin, David Caplan, Nathaniel Alpert & Scott Rauch. 1996. Localization of syntactic comprehension by positron emission tomography. *Brain and language* 52(3). 452–473. DOI: https://doi.org/10.1006/brln.1996.0024
- Swinney, David, Edgar Zurif, Penny Prather & Tracy Love. 1996. Neurological distribution of processing resources underlying language comprehension. *Journal of Cognitive Neuroscience* 8(2). 174–184. DOI: https://doi.org/10.1162/jocn.1996.8.2.174
- Tang, Gladys & Prudence Lau. 2012. In Pfau Roland, Markus Steinbach & Bencie Woll (eds.), *Coordination and Subordination*, 340–364. Berlin: De Gruyter.
- Traxler, Matthew J., Rihanna S. Williams, Shelley A. Blozis & Robin K. Morris. 2005. Working memory, animacy, and verb class in the processing of relative clauses. *Journal of Memory and Language* 53(2). 204–224. DOI: https://doi.org/10.1016/j.jml.2005.02.010
- Van Valin, Robert D. & David Wilkins. 1996. The case for "effector": Case roles, agents, and agency revisited. In M. Shibatani & S. A. Thompson (eds.), *Grammatical constructions: Their form and meaning*, 289–322. Oxford: Clarendon Press.
- Vasishth, Shravan & Richard L. Lewis. 2006. Argument-head distance and processing complexity: Explaining both locality and antilocality effects. *Language* 82(4). 767–794. DOI: https://doi. org/10.1353/lan.2006.0236
- Wilbur, Ronnie. 2017. Internally-headed relative clauses in sign languages. Glossa: a journal of general linguistics 2(1). 25. DOI: https://doi.org/10.5334/gjgl.183
- Wu, Fuyun, Elsi Kaiser & Elaine Andersen. 2009. The effect of classifiers in predicting Chinese relative clauses. In Proceedings of the Western Conference on Linguistics (WECOL). University of California Davis. Proceedings online: http://wecol. ucdavis. edu.
- Xiang, Ming, Brian Dillon, Matt Wagers, Fengqin Liu & Taomei Guo. 2014. Processing covert dependencies: an sat study on mandarin wh-in-situ questions. *Journal of East Asian Linguistics* 23. 207–232. DOI: https://doi.org/10.1007/s10831-013-9115-1
- Xiang, Ming, Sui Ping Wang & Yan Lin Cui. 2015. Constructing covert dependencies—the case of mandarin wh-in-situ dependency. *Journal of Memory and Language* 84. 139–166. DOI: https://doi. org/10.1016/j.jml.2015.05.006

Assessment tools

- Aristodemo, Valentina & Naama Friedmann. 2019. COGONEOUTLSF- *Cognitive odd one out test for LSF*. Handle to be attributed. Available under request at *https://www.sign-hub.eu/assessment/lsf*.
- Giustolisi, Beatrice & Naama Friedmann. 2019. COGONEOUTLIS *Cognitive odd one out test for LIS*. Handle to be attributed. Available under request at *https://www.sign-hub.eu/assessment/lis*.
- Giustolisi, Beatrice, Sala Rita, Caterina Donati, Naama Friedmann & Carlo Cecchetto. 2019. SYNTRELLIS-Relative clauses comprehension task in LIS. Handle to be attributed. Available under request at https://www.sign-hub.eu/assessment/lis.
- Hauser, Charlotte, Carlo Cecchetto, Naama Friedmann, Caterina Donati. 2019. SYNTRELLSF- *Relative clauses comprehension task in LSF*. Handle to be attributed. Available under request at *https://www.sign-hub.eu/assessment/lsf*.
- Zorzi, Giorgia, Jordina Sánchez Amat, Carlo Cecchetto, Caterina Donati, Naama Friedmann, Josep Quer. 2019. SYNTRELLSC- *Relative clauses comprehension task in LSC*. Handle to be attributed. Available under request at *https://www.sign-hub.eu/assessment/lsc*.
- Zorzi, Giorgia, Jordina Sánchez Amat & Naama Friedmann. 2019. COGONEOUTLSC-Cognitive odd one out test for LSC. Handle to be attributed. Available under request at https://www.sign-hub.eu/assessment/lsc.

Hauser et al. Glossa: a journal of general linguistics DOI: 10.5334/gjgl.1454

TO CITE THIS ARTICLE:

Hauser, Charlotte, Giorgia Zorzi, Valentina Aristodemo, Beatrice Giustolisi, Doriane Gras, Rita Sala, Jordina Sánchez Amat, Carlo Cecchetto and Caterina Donati. 2021. Asymmetries in relative clause comprehension in three European sign languages. *Glossa: a journal of general linguistics* 6(1): 72. 1–36. DOI: https://doi.org/10.5334/ gjgl.1454

Submitted: 26 September 2020 Accepted: 22 March 2021 Published: 31 May 2021

COPYRIGHT:

© 2021 The Author(s). This is an open-access article distributed under the terms of the Creative Commons Attribution 4.0 International License (CC-BY 4.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited. See http:// creativecommons.org/ licenses/by/4.0/.

Glossa: a journal of general linguistics is a peer-reviewed open access journal published by Ubiquity Press.

]u[👌